

Contents lists available at ScienceDirect

Journal of International Financial Markets, Institutions & Money



Fall of dwarfs: micro and macroeconomic determinants of the disappearance of European small banks

Federica Poli^{a,*,1}, Simone Rossi^b, Mariarosa Borroni^a

^a Department of Economic Sciences and Business Administration, Università Cattolica del Sacro Cuore, Milan, Italy
^b Department of Economic and Social Sciences, Università Cattolica del Sacro Cuore, Piacenza, Italy

| ARTICLE INFO | A B S T R A C T |
|---------------------------------|--|
| JEL codes: G21 G28 G17 | Based on a wide sample of banks headquartered in 27 European countries over the period 2005–2022, this paper tests the influence that microeconomic and macroeconomic variables have on the probability of small banks exiting the market, evaluating the predictive power of the explanatory models employed. Our approach to the determinants of small banks' exit proves that even the macroeconomic and socio-demographic reference context can have a predictive effectiveness similar to that of the accounting variables, especially when contagion effects at the local level are taken into account |

1. Introduction and literature review

After the outbreak of the global financial crisis that followed the collapse of Lehman Brothers, the major world economies witnessed a significant increase in regulatory and supervisory activities. The purpose of this action was mainly to contain and prevent the distress of systemically important subjects (so-called too-big-to-fail banks), and the potentially related phenomena of transmission of financial instability at a global level. Within the European Union, alongside the strengthening of banks' capital requirements determined by the introduction of the Basel III agreements, the ambitious project of the Banking Union codified new rules for the supervision and crisis management of so-called significant entities. Overall, attention was therefore mainly directed towards intermediaries deemed too big to fail, neglecting (or, from a less radical viewpoint, paying less attention to) small and medium-sized banks. For these, the supervision and crisis management safeguards that were already in force have been maintained, with a significant degree of national discretion. This asymmetric treatment was the natural outcome of the regulators' view, which judged smaller independent banks to be entities incapable of generating systemic impacts in the event of a crisis; an assumption which has proved to be incorrect (Enria 2021). In fact, failures of small banks in the United States and Europe have shown that the distress of these banks can generate negative spillovers which exceed the expectations of the regulators. However, even in the empirical literature, the effort dedicated to investigating small banking entities appears very slight and is often focused on the US market, where community banks play an important role, especially in rural areas, and have been less impacted by the consolidation process that has taken place over the last decade (Federal Deposit Insurance Corporation 2020).

In many European countries, small-sized banks still represent a very significant share of the banking institutions: less significant

https://doi.org/10.1016/j.intfin.2024.102042

Received 29 November 2023; Accepted 4 August 2024

Available online 8 August 2024

^{*} Corresponding author at: Department of Economic Sciences and Business Administration, Università Cattolica del Sacro Cuore, Largo Gemelli 1, 20123 Milano, Italy.

E-mail address: federica.poli@unicatt.it (F. Poli).

¹ Orcid: 0000-0002-1660-4971.

^{1042-4431/© 2024} The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

institutions (LSIs) accounted for approximately 95 % of the total banks at the end of 2021 (European Central Bank 2022). At the same date, the banks included in our dataset (which hold total assets not exceeding 5 billion euros thus being a subsample of LSIs) account for 49.8 % of the total number and 8.9 % of the total assets of all credit institutions in EU27 countries (our elaborations on Bankscope/BankFocus, European Central Bank and European Banking Federation data).

Despite a widespread consolidation process having taken place in the banking sector, small-sized banks continue to exist and, if well managed, where their market share is noteworthy, they also significantly reduce the likelihood of episodes of economic recession occurring in their area (Chiorazzo et al. 2018), as well as the duration of such episodes (Petach et al. 2021; Scott Langford and Feldman 2022), contrary to what has been found for large banks (Calice and Gam 2023). The geographical concentration of small banks' operations may expose them to the risk of local economic downturns, but situations of this kind are not systematically reflected in their performance (Yeager 2004). In essence: "small is not necessarily ugly", i.e., synonymous with greater risks and negative performance in the long run. Considerations of this kind are also shared by Barth et al. (2023) who analyze small US minority banks over a time horizon of almost twenty years and verify that their degree of risk and their level of profitability are similar to that of other banks. Berger and Bouwman (2013) also obtain analogous findings: high capital endowment, stable funding, a good level of profitability, and higher insulation from downturns in capital markets are always trump cards for small banks, but even more so in times of crisis. Although their business model is quite traditional (the collection of deposits and financing of small and medium-sized local businesses), the emphasis placed on personalizing the service they provide and the extensive use of soft information in their customer relationships endows them with a competitive advantage vis-à-vis large banks (DeYoung et al. 2004), even in difficult periods such as the months characterized by the COVID-19 pandemic (Li and Strahan 2021).

Small banks remain a safeguard that allows the benefits of relationship banking to unfold; they play an active role in supporting the territory in which they are located, they promote values of mutuality and stakeholder focus (e.g., cooperative banks), and ultimately reduce the risk of financial exclusion and the development of "banking deserts". This explains, for instance, the numerous studies on small business lending and the contribution this makes to local and nationwide economic growth, especially in those countries characterized by the widespread presence of small banks and an economic environment in which SMEs play a key role (e.g., on the Italian case, see Bellucci et al. 2013; Coccorese and Shaffer 2021).

On the other hand, the literature that investigates small banks *per se* and, in particular, their ability to survive within the local environment is limited: where available, the analyses in question above all examine individual countries or limited geographical areas in which there is a high number of small banks, often following a specific business model, such as e.g. cooperative banking (Halling and Hayden 2008; Fiordelisi and Mare 2013; Chernykh 2014; Mare 2015). One contribution of this study is therefore to overcome these limitations by investigating the determinants of the disappearance of small banks in a supranational context, such as the European Union.

In the existing empirical literature that has investigated the determinants of bank failure or the disappearance of banks due to mergers and acquisitions (Worthington 2004; Beccalli and Frantz 2013), both micro and/or macroeconomic variables are introduced. The former are seen as capable of capturing the fundamental sources of ex-ante risk (Gonzalez-Hermosillo 1999; Maggiolini and Mistrulli 2005; Mare 2015) while the latter variables may improve the predictive power of the models based on bank-specific data only (Arena 2008; Männasoo and Mayes 2009; Ioannidis et al. 2010; Betz et al. 2014; Mare 2015). In fact, given the restricted operations of small banks, both in terms of financial activity and geographical reach, their sensitivity to the state of the local economy may prove to be high. A novel element of our research is that we enrich the set of macroeconomic variables with indicators that reflect at a socio-demographic level the human capital that exists in the regions in which small banks operate. The limited research available on the survival of small banks at the country level (e.g., Ostergaard et al. 2016 for Norway) shows that the quality of local human capital (or the lack thereof) strengthens/weakens the resilience of small banks.

Additionally, based on the work of Hardy (1998) and more recently, Poghosyan and Čihak (2011), we also run separate estimations to test the relative importance of other potential sources of bank exit, such as contagion effects between small banks located in limited geographical areas. It is worth noting that at the regional level, the failure of small banks may more easily create rumors and, potentially, bank-run phenomena. As far as we know, this is the first research to examine the effects of contagion among small banks, and specifically at a European level.

Motivated by these research outcomes, we therefore aim to test, first separately and then jointly, the influence that typicallyemployed micro and macroeconomic variables, socio-demographic factors and contagion effects have on the probability of disappearance of small banks headquartered in 27 European countries over the period 2005–2022 and to evaluate the predictive power of such models. We use a discrete-time representation of a continuous time proportional hazards model, the complementary log–log model (cloglog), which is frequently used when the probability of an event is either very small or very large and survival occurs in continuous time, but spell length is observed only at intervals, as is the case for bank exits recorded on an annual basis in our sample (Männasoo and Mayes 2009; Mare 2015).

Our research contributes in several ways to the rather sparse literature on the determinants of the disappearance of small European banks. In the first instance, it confirms the role of size, capitalization, riskiness, and operational efficiency, in keeping with what has been found in other studies that are less territorially and longitudinally broad. Our approach to the determinants of small banks' disappearance proves that, for these banks, the macroeconomic and socio-demographic reference context can have a predictive effectiveness similar to that of the accounting variables, especially when contagion effects at a local level are taken into account.

Our study is one of the very few works to have considered the contagion effect at the local level, tested it, and identified its significance in explaining small banks' exit from the market (later in the text the terms 'disappearance' and 'exit' will be used interchangeably). Our evidence also demonstrates that the socio-demographic context in which dwarf banks operate deserves to be taken into due consideration given the strength of the relationship that exists between these intermediaries and their clients (Ostergaard et al.

2016).

Further, we account for the role played by monetary policy in explaining the pattern of small bank disappearance, contributing to the scientific debate and empirical evidence linked to the recent period of low or even negative rates.

The paper is organized as follows: section 2 introduces the sample and the methodology employed to analyze the available dataset; section 3 shows and discusses our results and includes several robustness checks; finally, section 4 presents policy implications and section 5 concludes.

2. Methodology, variables, and descriptive statistics

2.1. Empirical methodology and identification of target variables

To investigate the role of micro and macro determinants of the exit of small European banks (i.e. with total assets not exceeding 5 billion euros: more details are provided in section 2.3) over the period 2005–2022, we collected two sets of data, respectively reflecting accounting-based bank-specific information from annual financial statements provided by BankScope and BankFocus, and macro-economic and socio-demographic information from Eurostat, the International Monetary Fund, and the European Central Bank. The macroeconomic and sociodemographic explanatory variables are collected both at the regional level (NUTS 2 level) and at the national level, where regional data is unavailable or where national data can be validly used in place of regional data. According to Eurostat, the NUTS classification (acronym of "Nomenclature of Territorial Units for Statistics") is a system which allows us to divide the territory of the European Union to collect statistics, analyze the socio-economic conditions of the different regions and frame regional policies. NUTS 2 level includes regions with an average minimum and maximum population threshold of respectively 800,000 and 3,000,000.

The preference for using macroeconomic and sociodemographic data at the regional level responds to the specific need to gain a better understanding of the conditions of the economic and social contexts in which small banks operate, given their limited ability to exploit the effects of geographical diversification.

The variables used in this empirical study were selected based on the literature that has so far investigated both the determinants of banks' disappearance, and the channels through which the survival of banks can be put at risk (for example at the bank balance sheet level and/or concerning the macroeconomic context of reference). Our analysis focuses on the near-term vulnerability of banks, identifying and evaluating potential idiosyncratic influence factors as well as regional and national structural weaknesses that can affect bank exits and incentives to screen and monitor risks.

In the empirical literature to date, different estimation approaches have been used to identify the determinants of bank exit from the market due to failure, distress, and/or acquisitions. For instance, Cleary and Hebb (2016) use multivariate discriminant analysis (MDA) to examine the failures of 132 U.S. banks over the 2002–2009 period. Wheelock and Wilson (2000) analyze the determinants of individual bank failures and acquisitions in the United States from 1984 to 1993 and use Cox (1972) proportional hazards (PH) models with time-varying covariates. Similarly, Cox et al. (2017) employ a set of Cox PH models to ascertain those financial ratios that distinguish US bank failures from 2005 to 2010; the same approach is used by Pappas et al. (2017) for a sample of 421 banks in 20 Middle and Far Eastern countries from 1995 to 2010. The fundamental assumption in the Cox model is that the hazards are proportional, which means that the relative hazard remains constant over time with different predictor or covariate levels. Additionally, the baseline hazard is an unspecified function, which makes it a semi-parametric model.

Logit and probit estimation models are also employed to analyze the determinants of bank exit. Estrella et al. (2000) estimate the likelihood of failure using cross-sectional logit regression on a sample of U.S. banks during the 1988–93 period to examine the relationship between bank failure and regulatory and non-regulatory capital ratios. Poghosyan and Čihak (2011) use a logistic probability model for a sample of European banks in the EU-25 countries from 1996–2007. Betz et al. (2014) focus on large EU banks which may hinder systemic stability from 2000 to 2013; they apply a logit model approach, analogously to other authors such as Cole and White (2012), Lo Duca and Peltonen (2013), and Sarlin and Peltonen (2013). Within the empirical literature, pooled logit models prevail over panel-based ones due to the relatively small number of bank crises at the country level (Betz et al. 2014). More recently, Chiorazzo et al. (2018) employ a pooled probit approach with time-fixed effects on a sample of small US commercial banks from 1997 to 2012. They argue that probit and logit models with firm-fixed effects do not produce consistent parameter estimates, and that those with random effects impose a distribution on the unobservable firm effects and require them to be independent of the other regressors (Wooldridge 2010).

Both logit and probit apply a symmetric link function to a linear combination of covariates. These symmetric link functions are found to have some limitations when the probability of the binomial response approaches 0 and 1 at different rates (as a function of covariate) (Chen et al. 1999). Asymmetric (or skewed) links may be more appropriate than a symmetric link function in unbalanced data (Czado and Santner 1992; Chen et al. 1999). The complementary log–log model (cloglog) is an asymmetric link function that can be used for a discrete-time representation of continuous time proportional hazards. The complementary log–log model specification for the hazard regression is consistent with a continuous time model and interval-censored survival time data (Jenkins 2005). The cloglog belongs to the discrete-time functional specifications applied when survival occurs in continuous time, but spell length is observed only at intervals as in our study, where bank exits are recorded annually. Guo (1993) observes that time-varying covariates offer an opportunity to examine the relationship between the probability of failure (or exit) and the changing conditions under which the exit happens. Equivalence between interval-censored discrete-time and continuous-time models with the proportional hazards assumption is shown by Prentice and Gloeckler (1978). Therefore, it is possible to transform the coefficients of this analysis into hazard ratios. The cloglog model yields estimates of the impact of the indicators on the conditional probability of bank disappearance, which means that we obtain probabilities of bank exit, conditional on surviving to a certain point in time. Each regression coefficient summarizes the

effect on the hazard of absolute changes in the corresponding covariates with coefficients not varying with survival time.

Männasoo and Mayes (2009) use a discrete complementary log–log (cloglog) model as an estimation approach of banks' distress for a sample of nearly 600 banks in Eastern European transition countries over the years 1995–2004. The same method is used by Fiordelisi and Mare (2013), Chiaramonte et al. (2015), and Mare (2015) respectively on a sample of Italian cooperative banks during 1997 and 2009, of European banks from 12 countries over the period 2001–2011, and of Italian cooperative banks between 1993 and 2011. Based on these findings, we employ a cloglog model for our sample of small European banks from 2005 to 2022.

The cloglog with time-varying covariates assumes the following general form:

$$\Pr(y_j=1|X) = 1 - \exp\{-\exp(\beta X + \gamma_j)\}$$

where Pr is the probability that a bank exits the market in a given time interval j, X is the vector of the time-varying explanatory variables, lagged by one year, and γ_j summarizes differences in values of the integrated hazard function, and can be specified either by temporal dummy variables or by different parametric functional forms (Jenkins 2005). To deal with problems of time dependency that arise when using these models, we include the vector of temporal dummy variables for each spell year at risk of exit (Jenkins 2005; Chiaramonte et al. 2015). Additionally, we use robust standard errors clustered on the unit of analysis. The target and explanatory variables used in this study are outlined in Table 1.

Following Mare (2015), we develop a three-stage estimation framework in which the selected sets of micro and macro explanatory variables are estimated separately and then jointly. In our model, the dependent binary variable is equal to 1 for banks that did not survive over time due to both "financial and strategic exit" (as in Chiorazzo et al. 2018). In further estimates, we run separate estimates considering only the individual types of exits mentioned above.

We classify as cases of financial exit those banks that disappeared due to bankruptcy or liquidation, or that – according to the classification of bank status available in Bankscope and BankFocus – were dissolved without any information on the underlying reasons. We also included in this group of exits those banks that disappeared due to mergers or takeovers which, in the absence of specific information, could have been aimed at rescuing troubled banks (Arena 2008). In the latter case, mergers/takeovers of small banks would represent a market solution to a situation in which a bank found itself in a possible condition of financial distress. Conversely, a bank involved in a business aggregation could be healthy *per se*, at least in the short run, but see a strategic option in the merger/acquisition transaction to avoid, for example, future fragilities due to more accentuated competitive local arenas (strategic exit).

Within the empirical literature, an identification strategy commonly used to define distressed and then merged banks (e.g., Gonzales-Hermosillo 1999; Betz et al. 2014) is the coverage ratio (computed as capital equity and loan reserves minus non-performing loans to total assets). Banks dissolved due to mergers or takeovers are classified as distressed or, equivalently, in a state of financial exit if their coverage ratio is less than zero before the merger/takeover. With it being impossible to use, as in Betz et al. (2014), an indicator such as the coverage ratio due to the limited availability of data on impaired loans in our sample, we relied on the Z-score index as an alternative and widely employed identification tool. The Z-score is an accounting-based indicator of banks' distance to default (Poghosyan and Čihak 2011; Vazquez and Federico 2012; Chiaramonte et al. 2015). Its empirical attractiveness rests on the fact that it does not require strong assumptions about the distribution of returns on assets (Roy 1952; Boyd and Graham 1986; Strobel 2011), other than the ability to use accounting data – which are generally widely available – for its computation. We calculated individual bank Z-scores as the sum of the pre-tax or gross return on assets (ROA) and the equity to total assets ratio over the three-year rolling standard deviation of pre-tax ROA (Chiaramonte et al. 2015). We used the gross ROA since taxation on bank profits is set at different rates in the various countries in our sample, thus affecting net profitability. Banks dissolved by mergers/takeovers are therefore classified as being in a state of financial distress if their annual Z-score lies in the first quartile of the sample distribution in the year preceding their exit. By contrast, those banks involved in business combinations that have a higher Z-score are classified as disappeared for strategic purposes.

2.2. Explanatory variables

Following Betz et al. (2014), we employ two sets of explanatory variables: those expressing the bank-specific vulnerability and related to the individual bank's financial statement, and those representing sources of local and country-specific vulnerability factors, e.g., macro-economic imbalances and socio-demographic traits that could affect the disciplinary power of the market over time.

As regards the bank-specific independent variables, we use the size of the bank (*Size*), approximated by the natural logarithm of total assets (in millions of euros), to capture possible scale effects (i.e., diversification, managerial know-how, external monitoring, etc.). Demsetz and Strahan (1997) established that increases in banks' total assets reduce firm-specific risk and are positively related to diversification. Similarly, Stiroh (2004) and Mercieca et al. (2007) stated that larger banks take fewer risks as bank size increases. Hakenes and Schnabel (2011) provided evidence that the implementation of the Basel II regulations drove smaller banks to take on more risk compared to larger banks because the latter have the option to choose between the Standardized and Internal Ratings Approaches. On these grounds, we hypothesize a negative relationship between the size and probability of exit of the banks in our sample.

An individual bank's capacity to absorb losses is approximated by the *Capital* variable, calculated as the ratio between equity and total assets as a percentage. Capital buffers lower the banks' likelihood of default (Shim 2013) and risks (Barra and Ruggiero 2021). Therefore, we assume a negative relationship between the level of capitalization and the probability of exit of the banks in our sample.

The Funding gap variable aims to proxy primarily individual banks' exposure to liquidity risk. It is calculated as the ratio of net

Variables description. This table reports variable definitions, data sources, and representative studies.

| Variable | Definition and source | Source | Representative studies |
|-----------------------|---|--|---|
| Dependent Var | iables | | |
| Event | Binary variable taking a value of 1 if a bank entered a financial or strategic exit, 0 otherwise. | Own calculations using data from BankScope and BankFocus. | |
| Financial exit | Binary variable taking a value of 1 if a bank entered a financial exit, 0 otherwise. Financial exit occurs when a bank goes bankrupt or is liquidated or dissolved without any further information. This status also includes banks dissolved due to merger or take-over and whose annual Z-score is in the first quartile of the sample distribution in the year preceding the bank's | Own calculations using data from BankScope and BankFocus. | |
| Strategic exit | disappearance. Binary variable taking a value of 1 if a bank entered a strategic exit, 0 otherwise. Strategic exit occurs when a bank disappears due to a merger or take-over and whose annual Z-score is in the second or higher quartile of the sample distribution in the year preceding the bank's disappearance. | Own calculations using data from BankScope and BankFocus. | |
| Bank-specific v | ariables | | |
| Size | Natural logarithm of total assets (in millions of euros). | Own calculations using data from BankScope and BankFocus. | Berger and Bouwman (2013); Chernykh (2014); Mare (2015); Clark et al. (2018); Barth et al. (2023) |
| Capital | Equity is divided by total assets in percentage. | BankScope and BankFocus. | DeYoung et al. (2004); Chernykh (2014); Mare (2015); Chiorazzo et al. (2018); Barth et al. (2023) |
| Funding gap | Net customer loans are divided by customer deposits. | Own calculations using data from BankScope and BankFocus. | Mare (2015); Chiorazzo et al. (2018) |
| Credit risk | Loan loss provisions are divided by total assets in percentage. | Own calculations using data from BankScope and BankFocus. | Mare (2015); Contreras et al. (2023) |
| RAROA | Annual risk-adjusted return on assets (RAROA). The latter is computed as the 3-year moving average of the gross annual return on assets (ROA) over the corresponding moving standard deviation of ROA. | Own calculations using data from BankScope and BankFocus. | Mercieca et al. (2007); Turk Ariss (2010); Clark et al. (2018) |
| Cost to income | Operating expenses as a share of the sum of net- interest revenue and other operating income in percentage. | BankScope and BankFocus. | Maggiolini and Mistrulli (2005); Poghosyan and Čihak (2011); Liu et al. (2013); Mare (2015); Chiaramonte et al. (2015); Pappas et al. (2017) |
| Income div | Non-interest income as a share of the sum of net- interest revenue and other operating income in percentage. | BankScope and BankFocus. | Stiroh (2004); Stiroh and Rumble (2006); Chiaramonte et al. (2015); Pappas et al. (2017); Chiorazzo et al. (2018) |
| Macroeconomi | c variables | | |
| Monetary policy | Annual residuals from a regression of the 3-month interbank rate on national real GDP growth rate and inflation. | Own calculations from data collected from the European Central Bank (ECB), Eurostat, and International Monetary Fund (IMF). | Molyneux et al. (2019); Caselli et al. (2020) |
| Concentration | Annual national Herfindahl-Hirschman Index (HHI) divided by 100. | European Central Bank (ECB). | Beck et al. (2013) |
| Unemployment | Annual unemployment rate from 15 to 74 years old (NUTS 2 figures). | Eurostat | Kočenda and Iwasaki (2022); Liu et al. (2013) |
| Tertiary education | Annual percentage of the population aged 25–64 years with tertiary education (levels 5–8) (NUTS 2 figures). | Eurostat | OECD (2021) |
| Elderly population | Annual percentage of the population aged 65 and over to the total population (NUTS 2 figures). | Eurostat | McFadden (2008); OECD (2021) |
| Population density | Annual thousands of inhabitants per square kilometer (NUTS 2 figures). | Eurostat | Ostergaard et al. (2016) |
| Contagion | Dummy variable equal to 1 when another small bank in the same region (NUTS 2 level) disappeared in the last 12 months and 0 otherwise. | Own calculations using data from BankScope and BankFocus. | Poghosyan and Čihak (2011) |

Sample composition. This table shows the estimation sample distribution by bank status (active banks versus disappeared banks) and size (in millions of euros) for each European country. The sample period is from 2005 to 2022.

| | Active banks | | | | Disappeared banks | | | |
|---------------------------|--------------|------------|-------------|-------|-------------------|------------|-------------|-------|
| Country | <250 | 250 - 1000 | 1000 - 5000 | Total | <250 | 250 - 1000 | 1000 - 5000 | Total |
| Austria | 225 | 146 | 32 | 403 | 114 | 42 | 5 | 161 |
| Belgium | 1 | 7 | 8 | 16 | 1 | 4 | 5 | 10 |
| Bulgaria | 2 | 5 | 5 | 12 | 0 | 0 | 1 | 1 |
| Croatia | 4 | 7 | 3 | 14 | 2 | 0 | 1 | 3 |
| Cyprus | 12 | 8 | 4 | 24 | 0 | 1 | 1 | 2 |
| Czech Republic | 0 | 2 | 7 | 9 | 0 | 0 | 2 | 2 |
| Denmark | 18 | 13 | 10 | 41 | 2 | 2 | 4 | 8 |
| Estonia | 3 | 1 | 1 | 5 | 1 | 0 | 0 | 1 |
| Finland | 78 | 34 | 16 | 128 | 33 | 4 | 3 | 40 |
| France | 13 | 27 | 31 | 71 | 7 | 10 | 10 | 27 |
| Germany | 381 | 442 | 342 | 1,165 | 229 | 214 | 47 | 490 |
| Greece | 0 | 2 | 2 | 4 | 0 | 1 | 1 | 2 |
| Hungary | 2 | 7 | 3 | 12 | 1 | 0 | 2 | 3 |
| Ireland | 9 | 0 | 0 | 9 | 0 | 0 | 0 | 0 |
| Italy | 133 | 122 | 58 | 313 | 140 | 81 | 52 | 273 |
| Latvia | 6 | 1 | 5 | 12 | 1 | 1 | 1 | 3 |
| Lithuania | 0 | 1 | 2 | 3 | 1 | 1 | 0 | 2 |
| Luxembourg | 5 | 15 | 13 | 33 | 9 | 6 | 7 | 22 |
| Malta | 4 | 3 | 2 | 9 | 0 | 1 | 0 | 1 |
| Netherlands | 0 | 1 | 7 | 8 | 0 | 1 | 0 | 1 |
| Poland | 108 | 5 | 8 | 121 | 3 | 1 | 1 | 5 |
| Portugal | 68 | 19 | 6 | 93 | 9 | 1 | 2 | 12 |
| Romania | 1 | 5 | 6 | 12 | 1 | 1 | 0 | 2 |
| Slovak Republic | 1 | 2 | 2 | 5 | 0 | 1 | 3 | 4 |
| Slovenia | 0 | 2 | 8 | 10 | 0 | 1 | 5 | 6 |
| Spain | 34 | 26 | 22 | 82 | 8 | 4 | 11 | 23 |
| Sweden | 21 | 31 | 18 | 70 | 0 | 0 | 0 | 0 |
| Total | 1,129 | 934 | 621 | 2,684 | 562 | 378 | 164 | 1,104 |
| Of which | | | | | | | | |
| Commercial | 129 | 173 | 185 | 487 | 42 | 52 | 99 | 193 |
| Cooperatives | 932 | 611 | 179 | 1,722 | 512 | 270 | 43 | 825 |
| Savings | 68 | 150 | 257 | 475 | 8 | 56 | 22 | 86 |
| Financial exit estimation | 1,092 | 917 | 614 | 2,623 | 176 | 130 | 92 | 398 |
| Strategic exit estimation | 1,114 | 919 | 601 | 2,634 | 386 | 248 | 72 | 706 |
| 2005–2012 | 827 | 922 | 564 | 2,313 | 101 | 77 | 45 | 223 |
| 2013–2022 | 1,096 | 914 | 610 | 2,620 | 461 | 301 | 119 | 881 |

customer loans to customer deposits (e.g., Mare 2015; Chiorazzo et al. 2018). Given that deposits are usually considered a more stable funding source than the interbank market or securities funding, a lower *Funding gap* is expected to be negatively associated with the probability of bank disappearance. An increasing *Funding gap* reveals that those banks finance their illiquid assets, such as customer loans, with sources other than deposits, which substantially increases their liquidity risk. Indeed, a higher *Funding gap* increases the bank's dependence on so-called market-based financing which carries comparatively higher costs than those associated with customer deposits. Thus, the *Funding gap* may also proxy for the exposure to interest rate risk.

We proxy asset quality through the ratio of loan loss provisions to total assets as a percentage (*Credit risk*). A higher proportion of loan loss provisions to total assets is expected to increase the probability of disappearance, for instance, due to distress. However, the effect of provisioning is potentially ambiguous: whereas higher loan loss provisions could proxy for higher expected losses, they may be an accounting device for earning and capital management (e.g., Kanagaretnam et al. 2004; Jin et al. 2018; Tran et al. 2020; Contreras et al. 2023).

We look at individual bank risk and follow Mercieca et al. (2007), Turk Ariss (2010), and Clark et al. (2018) in computing a riskadjusted performance measure, the risk-adjusted return on assets (*RAROA*). The latter is calculated as the three-year moving average of the annual gross ROA divided by the corresponding moving standard deviation of gross ROA. Higher values of *RAROA* indicate higher profitability per unit of earning volatility, thus lowering exposure to the risk of exit.

The managerial quality of the bank is approximated by the cost to income ratio, *Cost to income*, (e.g., Poghosyan and Čihak 2011; Mare 2015; Chiaramonte et al. 2015; Pappas et al. 2017). This is calculated as operating expenses (e.g., staff and administrative costs) divided by the sum of net interest revenue and other operating income. Higher values denote higher managerial inefficiency, thus leading to a decrease in the probability of bank survival.

Finally, to account for the level of income diversification we include a variable (*Income div*) calculated as non-interest income over interest and non-interest operating revenues. The traditional business model of small banks relies mainly on borrowing and lending; therefore, we can expect a prevalence of interest income over operating revenues (Chiorazzo et al. 2018). In a classical à la Markowitz framework, the emergence of fee-based streams of revenues can generate diversification benefits; however, as stated by Stiroh and Rumble (2006), when the high volatility characterizing these incomes offsets the beneficial effect of covariance, the final effect on the

Descriptive Statistics. This table reports summary statistics on micro and macroeconomic explanatory variables as described in Table 1 for active and disappeared banks over the period 2005–2022. The column "Diff in means" reports the value of the mean-comparison tests in which the null hypothesis is that the means of the two groups (active and disappeared banks) are equal. The superscripts ***, **, and * denote coefficients statistically different from zero at the 1%, 5%, and 10% levels, respectively, in two-tailed tests.

| | Active banks | | Disappeared ban | ks | Diff in means |
|------------------------|--------------|---------|-----------------|---------|---------------|
| | Mean | St.Dev. | Mean | St.Dev. | |
| Size (Ln) | 6.266 | 1.216 | 5.777 | 1.148 | 0.488*** |
| Capital (%) | 9.908 | 5.280 | 8.985 | 4.536 | 0.924*** |
| Funding gap | 0.989 | 1.333 | 1.093 | 1.395 | -0.104*** |
| Credit risk (%) | 0.211 | 0.527 | 0.420 | 0.731 | -0.209*** |
| RAROA | 6.595 | 8.397 | 4.950 | 7.858 | 1.645*** |
| Cost to income (%) | 70.500 | 15.389 | 62.271 | 29.460 | 8.230*** |
| Income div (%) | 31.609 | 16.089 | 30.715 | 13.518 | 0.894*** |
| Monetary policy | -0.111 | 0.371 | -0.148 | 0.390 | 0.0370*** |
| Concentration | 4.835 | 4.254 | 4.094 | 3.299 | 0.741*** |
| Unemployment (%) | 6.045 | 4.412 | 6.417 | 4.119 | -0.372*** |
| Tertiary education (%) | 27.760 | 7.962 | 24.597 | 7.820 | 3.163*** |
| Elderly population (%) | 20.353 | 2.472 | 20.196 | 2.312 | 0.158*** |
| Population density | 0.337 | 0.635 | 0.365 | 0.758 | -0.0280** |
| Number of banks | 2,684 | | 1,104 | | |

overall profitability of a bank can be negative. Hence, in line with Chiaramonte et al. (2015) the expected sign of the coefficient associated with the variable *Income div* on bank disappearance is unclear.

To capture the effect of the macroeconomic and social context on the disappearance of small banks, several variables have been selected, preferably having a local dimension where possible, given the limited geographic operations of small banks and their strong dependence on local conditions.

To proxy the effects of monetary policy shocks on small bank exit we follow, among others, Caselli et al. (2020) and compute the Taylor rule residuals (Taylor 1993), since in our empirical set-up most of the countries are characterized by identical monetary policy rates but different economic conditions in terms of real GDP growth and inflation. The explanatory variable (Monetary Policy) is built using annual country data on the three-month interbank interest rate (%) collected from Eurostat and the European Central Bank regressed on the real GDP growth (%) from the International Monetary Fund, and the inflation rate (%) from the European Central Bank. Taylor rule residuals are estimated with a Panel Least Squares (PLS) regression for the eurozone economies and via a separate OLS regression for the other economies. Residuals depict the unexplained component of fluctuations in monetary policy rates (Brissimis et al. 2014) while accounting for cross-sectional variation in monetary conditions (Maddaloni and Peydró 2013). Positive (or negative) Taylor rule residuals denote a tight (or loose) monetary environment. According to the theoretical foundations of the risktaking channel (e.g., Borio and Zhu 2012; Angeloni et al. 2015), looser monetary conditions prompt banks to operate closer to default (Molyneux et al. 2019; Caselli et al. 2020) and vice versa. Given that small banks' profitability tends to depend strongly on the interest margin, we can expect more restrictive monetary policies to improve bank interest margins (Alessandri and Nelson 2015; Borio and Gambacorta 2017; Claessens et al. 2018) and therefore the resilience of these banks to distress phenomena that may also result in M&A transactions mainly aimed at rescuing troubled banks. However, it has been noted that looser monetary policy can also exert a positive influence on banks' profitability and stability. This happens through different channels, including lower borrowing rates, sounder overall economic outlook – which can reduce NPLs and enhance lending opportunities – and a growth in the value of assets, including government and corporate bonds (Demertzis and Wolff 2016; Altavilla et al. 2018; Lopez et al. 2020; Altavilla et al. 2022). Therefore, the expected effect of monetary policy on banks' disappearance is controversial and needs to be empirically determined.

To evaluate the role of competition on the fragility/stability of sample banks in line with the competition-fragility view (Marcus 1984; Keeley 1990; Diallo 2015) versus the competition-stability view (Boyd and De Nicolò 2005; De Nicolò and Lucchetta 2009; Schaeck et al. 2009), we employ the variable *Concentration*. Unfortunately, there is no valid measure of competition at the local level for our specific sample and therefore we use an indicator of concentration, such as the Herfindahl-Hirschman Index (HHI) at the country level. Despite the theoretical meaning of the competition-fragility view which identifies the increase in the degree of competition as the cause of banks' greater fragility (e.g., Beck et al. 2013), it seems useful to us to observe that higher degrees of concentration in the national banking system may expose small banks to greater competitive pressures from large banks in their local environment, which erode their profits and/or push them to take on higher levels of risk, thus ultimately leading to an increased probability of exit. Stakeholder-oriented banks, such as cooperative and savings banks, can however be less exposed to such pressures, given their not-for-profit orientation (e.g., Ayadi et al. 2009; Ayadi et al. 2010). On the other hand, as bank market power increases at the national level, the riskiness of the credit portfolios held by large banks may rise, thus leaving room for them to disengage from lending activity, especially from more opaque retail lending. This can favor the activity and survival of small banks if they are willing to make greater investments in collecting information, screening, and monitoring (Clark et al. 2018). The effect of the degree of national concentration on the disappearance of small banks is therefore uncertain.

Additionally, we consider the unemployment rate (*Unemployment*) at the regional level, which can impact economic growth at a local level and ultimately the creditworthiness of bank borrowers. The literature based on U.S. and European banks shows that various

Main estimates – full sample. This table shows in Panel A the complementary log–log model estimations obtained by regressing bank exit on the micro and macroeconomic lagged variables as described in Table 1. Reported coefficients are not exponentiated. The explanatory dummy variable Contagion is added in some specifications and is described in Table 1. The sample period is 2005–2022. All microeconomic variables are winsorized at the 1% of each tail. Temporal dummy variables are also included in the model. Robust standard errors clustered on the unit of analysis are reported in parentheses. The superscripts ****, ***, ***, and * denote coefficients statistically different from zero at the 0.1%, 1%, 5%, and 10% levels, respectively, in two-tailed tests. This table also displays in Panel B models' goodness of fit concerning the relationship between model predictions and actual events of bank exit. In-sample predictions are computed on the whole sample used to fit the models using a cut-off point equal to 1%, 2%, and 3%; out-of-sample predictions are calculated on a training sample cover the period 2005–2017 and a hold-out period covering the years 2018–2022. Further out-of-sample predictions are calculated on a training sample consisting of banks located in Austria, Germany, and Italy and a hold-out sample consisting of banks in the remaining countries. For both out-of-sample predictions, only those with a 3% cut-off are presented. Panel B also reports the ranking of disappeared banks using the estimated hazard rate, from the least to the most risky. The hazard rate is computed using the estimated coefficients for the variables in the model using different specifications for the banks in-sample over the whole period and out-of-samples. Following Hosser and Lemeshow (2000) we do not report any R-squared measure (such as the Nagelkerke's pseudo-R-squared) since the authors indicate that a true measure of fit is the one based strictly on a comparison of observed to predicted values from the fitted model.

| Panel A | | | | | | | |
|------------------------|---------------------------|----------------|---|-----------------|----------|----------------|----------------|
| | | Mo | odel 1 | Model 2 | Model 3 | Model 4 | Model 5 |
| L.Size | | -0.2 | 275**** | -0.284**** | | | -0.294**** |
| | | (0 | .027) | (0.028) | | | (0.029) |
| LCapital | | -0.0 |)52**** | -0.053**** | | | -0.056**** |
| - | | (0 | .009) | (0.010) | | | (0.010) |
| L.Funding gap | | 0.0 | 48*** | 0.043** | | | 0.0416* |
| | | (0 | .016) | (0.018) | | | (0.022) |
| L.Credit risk | | 0.45 | 58**** | 0.453**** | | | 0.417**** |
| | | (0 | .045) | (0.047) | | | (0.051) |
| L.RAROA | | _(| 0.008 | -0.008 | | | -0.007 |
| | | (0 | .005) | (0.005) | | | (0.005) |
| L.Cost to income | | 0.02 | 20**** | 0.020**** | | | 0.021**** |
| | | (0 | .002) | (0.002) | | | (0.002) |
| L.Income div | | 0.0 | 06*** | 0.005** | | | 0.005** |
| | | (0 | .002) | (0.002) | | | (0.002) |
| L.Monetary policy | | | | | -0.284** | -0.262 | -0.276* |
| | | | | | (0.117) | (0.191) | (0.159) |
| L.Concentration | | | | | 0.005 | 0.0178** | 0.018*** |
| | | | | | (0.008) | (0.008) | (0.009) |
| L.Unemployment | | | | | 0.019** | 0.037**** | 0.006 |
| T Thereicher Anneriter | | | | | (0.007) | (0.008) | (0.009) |
| L. Tertiary eaucation | | | | | -0.001 | -0.010^ | -0.005* |
| I Eldonho noncolation | | | | | (0.005) | (0.005) | (0.006) |
| L.Elderly population | | | | | 0.030** | 0.035** | 0.056^^ |
| I Donalation donaits | | | | | (0.014) | (0.017) | (0.018) |
| L.Population density | | | | | 0.079" | (0.048) | 0.093* |
| Contagion | | | | 1 60/**** | (0.044) | (0.046) | (0.030) |
| Contagion | | | | (0.0713) | | (0.072) | (0.073) |
| Voar dummies | | | Voc | (0.0713) Vec | Voc | (0.072) Vec | (0.073) Vec |
| No. of disappeared h | anke | 1 | 104 | 1 104 | 1 104 | 1 104 | 1 104 |
| No. of banks | AIIKS | 3 | 788 | 3 788 | 3 788 | 3 788 | 3 788 |
| Mean hazard of disan | neared banks | 5 | ,700 | 5,700 | 5,700 | 5,700 | 0.089 |
| Mean hazard | peureu burius | | | | | | 0.031 |
| No. of obs | | 35 | 5.033 | 35.033 | 35.033 | 35.033 | 35.033 |
| | | 0.0 | ,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,, | 00,000 | 00,000 | 00,000 | 00,000 |
| Panel B | M. J.1.1 | M- 1-10 | M- 1-1.0 | 34-1-14 | | M. 4.1 C | |
| | Model 1 | Model 2 | Model 3 | Model 4 | | Model 5 | |
| In-sample predictions | 1 % cut-off (%) | | | | | | |
| Sensitivity | 95.83 | 91.76 | 99.91 | 95.47 | | 91.49 | |
| Specificity | 9.94 | 32.65 | 0.34 | 18.06 | | 33.43 | |
| Corr. classified | 12.65 | 34.51 | 3.48 | 20.50 | | 35.26 | |
| ROC area | 52.89 | 62.20 | 50.13 | 56.77 | | 62.46 | |
| In-sample predictions | 2 % cut-off (%) | | | | | 60.44 | |
| Sensitivity | 84.51 | 79.80 | 86.32 | 74.82 | | 62.46 | |
| Specificity | 39.46 | 79.80 | 27.68 | 63.62 | | 61.38 | |
| Corr. classified | 40.88 | 62.09 | 29.52 | 63.98 | | 61.96 | |
| KUC area | 61.98 | 70.66 | 57.00 | 69.22 | | 70.50 | |
| in-sample predictions | 3 % CUT-Off (%) | 72.01 | 69.02 | 71.02 | | 79.64 | |
| Sensitivity | 62.49 | / 3.91 | D8.U3 | /1.83 | | / 3.04 | |
| Specificity | 03.48 62 E9 | 60.40 | 53.U8 E2 EE | 60.02 | | 09.01 60.74 | |
| COIL CLASSIFIED | 03.38 64.09 | 09.49 71.69 | 53.55 E2 EE | 70.20 | | 09.74 | |
| In comple disconnector | 04.98 Lhanke ranking h | 1.00 / 1.00 | 33.33 anand nata (04) | /0.39 | | /1.03 | |

In-sample disappeared banks ranking by decile of the hazard rate (%)

(continued on next page)

Table 4 (continued)

| Panel B | | | | | |
|-----------------------|------------------|---------------------|--------------------|---------------------|---|
| | Model 1 | Model 2 | Model 3 | Model 4 | Model 5 |
| 1–5 decile | 23.37 | 14.95 | 29.89 | 17.66 | 15.22 |
| 6 | 7.70 | 5.25 | 11.68 | 5.89 | 5.16 |
| 7 | 9.24 | 8.24 | 10.78 | 7.34 | 6.97 |
| 8 | 10.96 | 11.41 | 14.13 | 16.30 | 12.05 |
| 9 | 14.49 | 16.85 | 15.40 | 19.93 | 17.30 |
| 10 | 34.24 | 43.30 | 18.12 | 32.88 | 43.30 |
| Out-of-sample predict | ions 3 % cut-off | f (hold-out sample | 2018–2022) (%) |) | |
| Sensitivity | 33.56 | 60.50 | 0.00 | 70.73 | 59.13 |
| Specificity | 83.54 | 78.26 | 100.00 | 67.50 | 78.72 |
| Corr. classified | 81.80 | 77.64 | 96.51 | 67.61 | 78.04 |
| ROC area | 58.55 | 69.38 | 50.00 | 69.11 | 68.92 |
| Out-of-sample disapp | eared banks ran | king by decile of t | he hazard rate (h | old-out sample 2018 | 3–2022) (%) |
| 1–5 decile | 33.56 | 16.89 | 52.99 | 19.23 | 18.51 |
| 6 | 7.53 | 5.02 | 9.83 | 5.34 | 4.63 |
| 7 | 10.27 | 8.45 | 8.76 | 13.68 | 7.20 |
| 8 | 10.96 | 15.07 | 10.47 | 15.81 | 14.40 |
| 9 | 12.56 | 17.58 | 7.26 | 18.80 | 16.97 |
| 10 | 25.11 | 36.99 | 10.68 | 27.14 | 38.30 |
| Out-of-sample predict | ion 3 % cut-off | (hold-out sample | with countries oth | er than Austria, Ge | many, and Italy) (%) |
| Sensitivity | 55.49 | 59.34 | 83.71 | 70.45 | 52.78 |
| Specificity | 54.73 | 73.64 | 13.64 | 54.31 | 75.44 |
| Corr. classified | 54.76 | 73.21 | 15.68 | 54.78 | 74.76 |
| ROC area | 55.11 | 66.49 | 48.68 | 62.38 | 64.11 |
| Out-of-sample disapp | eared banks ran | king by decile of t | he hazard rate (h | old-out sample with | countries other than Austria, Germany, and Italy, (%) |
| 1–5 decile | 41.21 | 26.92 | 50.76 | 26.14 | 29.44 |
| 6 | 9.89 | 7.69 | 10.98 | 15.53 | 7.22 |
| 7 | 10.99 | 3.85 | 6.06 | 9.47 | 6.67 |
| 8 | 12.09 | 8.24 | 6.06 | 5.30 | 10.00 |
| 9 | 8.24 | 15.93 | 9.09 | 14.02 | 15.56 |
| 10 | 17.58 | 37.36 | 17.05 | 29.55 | 31.11 |

proxies for economic development (such as unemployment, stock market volatility, GDP growth, real estate investment, etc.) improve predictions of bank distress (Liu et al. 2013; Kočenda and Iwasaki 2022). We expect a positive relationship between *Unemployment* and bank exit.

In line with the analysis by Ostergaard et al. (2016) on the factors determining the survival of Norwegian savings banks, we include some variables that approximate local human capital and its ability to strengthen/weaken the resilience of small banks. Following Coleman (1988), we adopt the notion that human capital is created by changes in personnel that generate skills and capabilities that enable the bank to act in new ways, e.g., by promoting economic development, fostering monitoring of economic and financial activities, etc. Human capital increases with general training and knowledge and tends to depreciate with age (McFadden 2008; OECD 2021). To this end, we have selected two variables, which respectively approximate, at the local level, the prevalence in the population of advanced education qualifications (*Tertiary education*) and the proportion of older citizens aged 65 and over (*Elderly population*). Regarding the *Tertiary education* variable, we hypothesize that a more widespread distribution of higher education could favor the survival of small banks, especially if it benefits local economic development and improves the stakeholders' ability to monitor and discipline the management of financial intermediaries. Higher levels of educational attainment can also help stimulate the adoption of best practices in banking risk management, thus improving bank resilience. By contrast, we are led to hypothesize that as the share of the elderly population increases, the survival of small banks is more at risk, because, in the long run, the contributions that human capital can make to local development and the stability of the existing organizations may gradually decrease.

Moreover, following Ostergaard et al. (2016), we control for the thousands of inhabitants per square kilometer (*Population density*), which may exert divergent effects on bank survival. Since the supply of banking services in areas with low population density tends to raise banking costs, the survival of banks located in such areas may be more at risk (Dietsch and Lozano-Vivas 2000). On the other hand, increasing population density may, for instance, boost competitive pressures or help to spread contagion effects among local banks. Therefore, the expected effect of this covariate on small banks' disappearance is uncertain.

Finally, we control for another variable which is aimed at capturing contagion between banks, in line with the empirical contributions of Hardy (1998) and Poghosyan and Čihak (2011). We construct the variable *Contagion* as a dummy assuming a value of 1 when another small bank in the same region (NUTS 2 level) exited in the last 12 months due to both financial and strategic exit, and zero otherwise. Given how geographically limited the territories taken into consideration (NUTS 2 level) are, the exit of one or more small banks is more likely to occur in the event of adverse macroeconomic local and national shocks. To evaluate the contribution of the variable *Contagion* we will run separate estimations. It is worth noting that the disappearance of small banks in limited geographical areas may generate rumors more easily and give rise to bank-run phenomena that may trigger financial exits. On the other hand, the increase in cases of strategic exit at a local level can stimulate emulative behaviors among local small banks fearing for their future survival.

Estimates by bank size. This table shows in Panel A the complementary log-log model estimations for different bank sizes obtained by regressing bank exit on the micro and macroeconomic lagged variables as described in Table 1. Reported coefficients are not exponentiated. The sample period is 2005-2022. All microeconomic variables are winsorized at the 1% of each tail. Temporal dummy variables are also included in the model. Robust standard errors clustered on the unit of analysis are reported in parentheses. The superscripts ****, ***, ***, and * denote coefficients statistically different from zero at the 0.1%, 1%, 5%, and 10% levels, respectively, in two-tailed tests. This table also displays in Panel B models' goodness of fit concerning the relationship between model predictions and actual events of bank exit. In-sample predictions are computed on the whole sample used to fit the models using a cut-off point equal to 1%, 2%, and 3%; out-of-sample predictions are calculated on a training sample over the period 2005-2017 and a hold-out period covering the years 2018-2022. Panel B also reports the ranking of disappeared banks using the estimated hazard rate, from the least to the most risky. The hazard rate is computed using the estimated coefficients for the variables in the model using different specifications for the banks in-sample over the whole period and out-of-sample. Following Hosmer and Lemeshow (2000) we do not report any Rsquared measure (such as the Nagelkerke's pseudo-R-squared) since the authors indicate that a true measure of fit is the one based strictly on a comparison of observed to predicted values from the fitted model.

| Panel A | | | |
|----------------------------------|---|---------------------------------|---------------------------------|
| | Assets $< 250 \text{ (mln } \epsilon$) | $250 (mln \epsilon) <=$ | 1000 (mln ϵ) <=Assets |
| | | Assets $< 1000 \text{ (mln t)}$ | <= 5000 (mln t) |
| L.Size | -0.392**** | -0.362^{***} | 0.036 |
| | (0.083) | (0.139) | (0.177) |
| L.Capital | -0.051^{****} | -0.109^{****} | -0.006 |
| | (0.012) | (0.021) | (0.024) |
| L.Funding gap | 0.068* | -0.065 | 0.071** |
| | (0.040) | (0.041) | (0.028) |
| L.Credit risk | 0.358**** | 0.464**** | 0.195 |
| | (0.093) | (0.098) | (0.121) |
| L.RAROA | -0.003 | 0.002 | -0.053^{**} |
| | (0.008) | (0.007) | (0.022) |
| L.Cost to income | 0.026**** | 0.022**** | 0.012*** |
| | (0.003) | (0.004) | (0.005) |
| L.Income div | -0.003 | 0.003 | -0.002 |
| | (0.008) | (0.004) | (0.005) |
| L.Monetary policy | -0.518** | -0.227^{**} | -0.863^{**} |
| | (0.243) | (0.102) | (0.392) |
| L.Concentration | 0.011 | 0.012 | 0.046** |
| | (0.013) | (0.018) | (0.018) |
| L.Unemployment | -0.011 | 0.012* | 0.084**** |
| | (0.013) | (0.018) | (0.016) |
| L.Tertiary education | -0.023^{**} | 0.001 | -0.003 |
| | (0.010) | (0.010) | (0.012) |
| L.Elderly population | 0.090*** | 0.081*** | 0.041 |
| | (0.028) | (0.028) | (0.040) |
| L.Population density | 0.108 | 0.030 | 0.115* |
| | (0.163) | (0.087) | (0.067) |
| Contagion | 1.820**** | 1.750**** | 1.346**** |
| | (0.115) | (0.134) | (0.179) |
| Year dummies | Yes | Yes | Yes |
| No. of disappeared banks | 482 | 420 | 202 |
| No. of banks | 1,611 | 1,354 | 823 |
| Mean hazard of disappeared banks | 0.028 | 0.068 | 0.031 |
| Mean hazard | 0.013 | 0.028 | 0.016 |
| No. of obs. | 11,442 | 12,686 | 9,519 |
| Panel B | | | |

| | Assets $<250~(mln~\mbox{\pounds})$ | $\begin{array}{l} \text{250 (mln ε)} \\ <= \text{Assets} < 1000 (mln ε) \end{array}$ | 1000 (mln e) <=Assets <= 5000 (mln e) |
|---|------------------------------------|--|--|
| In-sample predictions 1 % cut-off (%) | | | |
| Sensitivity | 94.74 | 92.64 | 86.11 |
| Specificity | 24.78 | 42.90 | 50.31 |
| Corr. classified | 27.69 | 44.44 | 50.98 |
| ROC area | 59.76 | 67.77 | 68.21 |
| In-sample predictions 2 % cut-off (%) | | | |
| Sensitivity | 85.47 | 83.25 | 70.00 |
| Specificity | 55.02 | 65.79 | 72.12 |
| Corr. classified | 56.28 | 66.33 | 72.08 |
| ROC area | 70.25 | 74.52 | 71.06 |
| In-sample predictions 3 % cut-off (%) | | | |
| Sensitivity | 81.47 | 76.65 | 51.67 |
| Specificity | 64.78 | 72.43 | 83.45 |
| Corr. classified | 65.47 | 72.56 | 82.84 |
| ROC area | 73.12 | 74.54 | 67.56 |
| In-sample disappeared banks ranking by decile | of the hazard rate (%) | | |

ample disappeared banks ranking by decile of the hazard rate (%)

(continued on next page)

Table 5 (continued)

| Panel B | Assets $<250~(mln~\varepsilon)$ | $250 (\min f)$ | 1000 (mln ϵ) <=Assets |
|---|--|-----------------------------|---------------------------------|
| | | <= /issets < 1000 (iiiii c) | <= 5000 (iiiii e) |
| 1–5 decile | 13.26 | 9.90 | 13.89 |
| 6 | 4.21 | 4.57 | 6.11 |
| 7 | 8.42 | 7.87 | 8.89 |
| 8 | 10.53 | 11.93 | 15.56 |
| 9 | 19.37 | 19.80 | 17.22 |
| 10 | 44.21 | 45.94 | 38.33 |
| Out-of-sample predictions 3 % cut-off (hold-out | sample 2018–2022) (%) | | |
| Sensitivity | 47.81 | 75.17 | 36.47 |
| Specificity | 76.73 | 60.21 | 73.89 |
| Corr. classified | 75.23 | 60.68 | 73.13 |
| ROC area | 62.27 | 67.69 | 55.18 |
| Out-of-sample disappeared banks ranking by dee | cile of the hazard rate (hold-out sample 2 | 2018–2022) (%) | |
| 1–5 decile | 16.75 | 18.11 | 22.22 |
| 6 | 6.70 | 3.15 | 2.78 |
| 7 | 11.96 | 8.66 | 22.22 |
| 8 | 13.88 | 22.05 | 15.28 |
| 9 | 16.75 | 14.96 | 15.28 |
| 10 | 33.97 | 33.07 | 22.22 |

2.3. Sample and descriptive statistics

The sample consists of 3,788 small banks headquartered in 27 European countries (Table 2).

In our study, we classify as a "small bank" a credit institution with total assets not exceeding 5 billion euros, consistent with article 4 (1), point 145 of EU Regulation No 575/2013, which sets this threshold for small and non-complex entities. This definition has also been adopted by the European Banking Authority in line with its Guidelines on common procedures and methodologies for the Supervisory Review and Evaluation Process (SREP). The EU Single Resolution Board has also recently revised its resolution procedures (Guidance on the Liability Data Report 2023), defining organizations with total assets exceeding 5 billion euros as a "relevant legal entity". Therefore, credit institutions with total assets lower than the afore-mentioned threshold are not "relevant", even if they belong to a banking group that can be subject to resolution rules. To verify the factors that determine the disappearance of small banks of various sizes, three further sub-samples are identified, respectively comprising banks with assets of less than 250 million euros, those with assets of between 250 million and 1,000 million euros, and those with over 1,000 million but less than 5,000 million euros.

During the period of analysis, 2,684 banks were continuously active, and 1,104 disappeared, as a result of both financial and strategic exits.

Most of the banks in the sample (approximately three-quarters) are concentrated in just three countries: Austria, Italy, and Germany. The latter has 1,655 banks, almost half of the total. Banks with total assets of less than 1,000 million euros account for 79.3 % of the total number of banks. Since we report the post-estimation statistics, for some countries no cases of bank disappearance are available, even though they are present in the original sample. Among the non-surviving banks (29.1 % of the total sample), about 85 % have total assets below 1 billion euros.

The bottom lines of Table 2 show the composition of the sample for all the further sub-sample analyses and robustness checks performed.

As displayed in Table 3, the results relating to all the covariates are statistically different for active and disappeared banks: the former are bigger, more capitalized, have a lower funding gap and credit risks, higher risk-adjusted profitability, and follow a more diversified business, despite being less managerially efficient. The disappeared banks operate in environments with higher unemployment rates, less educated populations, a lower population age, less concentrated national banking systems, and higher Taylor rule residuals and population density.

3. Main findings

3.1. Baseline estimation results

Table 4 shows our model estimates for the entire sample of banks analyzed.

The first two columns contain the results of the analysis performed using only microeconomic variables as explanatory variables for our dependent variable (*Event*). The results show that several covariates make a significant contribution to determining the disappearance of small banks over the period examined.

The Model 1 column reports the results of the estimates using only balance sheet data from the surveyed banks as explanatory

Estimates by business model. This table shows in Panel A the complementary log–log model estimations for different bank business models obtained by regressing bank exit on the micro and macroeconomic lagged variables as described in Table 1. Reported coefficients are not exponentiated. The sample period is 2005–2022. All microeconomic variables are winsorized at the 1% of each tail. Temporal dummy variables are also included in the model. Robust standard errors clustered on the unit of analysis are reported in parentheses. The superscripts ****, ***, and * denote coefficients statistically different from zero at the 0.1%, 1%, 5%, and 10% levels, respectively, in two-tailed tests. This table also displays in Panel B models' goodness of fit concerning the relationship between model predictions and actual events of bank exit. In-sample predictions are computed on the whole sample used to fit the models using a cut-off point equal to 1%, 2%, and 3%; out-of-sample predictions are calculated on a training sample over the period 2005–2017 and a hold-out period covering the years 2018–2022. Panel B also reports the ranking of disappeared banks using the estimated hazard rate, from the least to the most risky. The hazard rate is computed using the estimated coefficients for the variables in the model using different specifications for the banks in-sample over the whole period and out-of-sample. Following Hosmer and Lemeshow (2000) we do not report any R-squared measure (such as the Nagelkerke's pseudo-R-squared) since the authors indicate that a true measure of fit is the one based strictly on a comparison of observed to predicted values from the fitted model.

| Panel A | | | |
|----------------------------------|------------------|-------------------|---------------|
| | Commercial banks | Cooperative banks | Savings banks |
| L.Size | 0.109 | -0.338**** | -0.546**** |
| | (0.072) | (0.038) | (0.116) |
| L.Capital | -0.013 | -0.062^{****} | -0.150*** |
| - | (0.011) | (0.0139) | (0.055) |
| L.Funding gap | 0.021 | 0.031 | 0.062 |
| | (0.022) | (0.078) | (0.103) |
| L.Credit risk | 0.162* | 0.415**** | 0.913** |
| | (0.087) | (0.079) | (0.396) |
| L.RAROA | -0.081^{****} | 0.002 | -0.011 |
| | (0.025) | (0.005) | (0.025) |
| L.Cost to income | 0.007** | 0.030**** | 0.045**** |
| | (0.003) | (0.004) | (0.010) |
| L.Income div | -0.0006 | 0.009** | 0.007 |
| | (0.003) | (0.003) | (0.012) |
| L.Monetary policy | -0.156 | -0.423* | -1.020 |
| | (0.176) | (0.225) | (0.971) |
| L.Concentration | 0.034** | 0.003 | 0.0341 |
| | (0.015) | (0.011) | (0.032) |
| L.Unemployment | 0.043** | -0.011 | 0.078 |
| | (0.018) | (0.012) | (0.052) |
| L.Tertiary education | -0.019 | 0.0002 | -0.004 |
| | (0.012) | (0.007) | (0.031) |
| L.Elderly population | 0.060 | 0.061*** | 0.224*** |
| | (0.037) | (0.021) | (0.070) |
| L.Population density | 0.052 | 0.203* | 0.283* |
| | (0.060) | (0.110) | (0.171) |
| Contagion | 1.383**** | 1.671**** | 2.144**** |
| | (0.171) | (0.089) | (0.332) |
| Year dummies | Yes | Yes | Yes |
| No. of disappeared banks | 193 | 825 | 86 |
| No. of banks | 680 | 2,547 | 561 |
| Mean hazard of disappeared banks | 0.071 | 0.034 | 0.059 |
| Mean hazard | 0.040 | 0.014 | 0.012 |
| No. of obs | 4,700 | 24,076 | 6,232 |

| T unici 2 | | | |
|-------------------------------------|-------------------------------------|-------------------|---------------|
| | Commercial banks | Cooperative banks | Savings banks |
| In-sample predictions 1 % cut-off (| (%) | | |
| Sensitivity | 98.96 | 93.33 | 82.56 |
| Specificity | 15.24 | 33.88 | 70.71 |
| Corr. classified | 18.68 | 35.92 | 70.88 |
| ROC area | 57.10 | 63.61 | 76.64 |
| In-sample predictions 2 % cut-off (| (%) | | |
| Sensitivity | 89.64 | 82.79 | 70.93 |
| Specificity | 42.22 | 59.05 | 84.23 |
| Corr. classified | 44.17 | 59.86 | 84.05 |
| ROC area | 65.93 | 70.92 | 77.58 |
| In-sample predictions 3 % cut-off (| (%) | | |
| Sensitivity | 77.72 | 77.70 | 60.47 |
| Specificity | 59.68 | 67.98 | 89.77 |
| Corr. classified | 60.43 | 68.31 | 89.36 |
| ROC area | 68.70 | 72.84 | 75.12 |
| In-sample disappeared banks ranki | ng by decile of the hazard rate (%) | | |
| 1–5 decile | 13.47 | 13.58 | 9.30 |
| 6 | 11.92 | 4.85 | 5.81 |
| | | | |

(continued on next page)

Table 6 (continued)

| Panel B | | | |
|----------------------------------|--|------------------------------|---------------|
| | Commercial banks | Cooperative banks | Savings banks |
| 7 | 8.29 | 7.27 | 2.33 |
| 8 | 13.99 | 11.52 | 8.14 |
| 9 | 16.06 | 16.97 | 16.28 |
| 10 | 36.27 | 45.82 | 58.14 |
| Out-of-sample predictions 3 % cu | ut-off (hold-out sample 2018–2022) (%) | | |
| Sensitivity | 80.65 | 43.38 | 68.75 |
| Specificity | 44.74 | 84.16 | 78.58 |
| Corr. classified | 46.53 | 82.57 | 78.44 |
| ROC area | 62.69 | 63.77 | 73.67 |
| Out-of-sample disappeared banks | ranking by decile of the hazard rate (ho | ld-out sample 2018–2022) (%) | |
| 1–5 | 22.39 | 17.42 | 42.90 |
| 6 | 13.43 | 4.80 | 28.60 |
| 7 | 7.46 | 8.41 | 14.30 |
| 8 | 16.42 | 13.51 | 0.00 |
| 9 | 10.45 | 19.22 | 0.00 |
| 10 | 29.85 | 36.64 | 14.30 |

variables. Reported coefficients are not exponentiated to obtain hazard ratios.² All covariates, with the exception of *RAROA*, appear significant, albeit with different signs. Larger (Maggiolini and Mistrulli 2005; Mare 2015) and well-capitalized banks (Betz et al. 2014; Mare 2015) are less exposed to market exit (measured by the variable *Event*). In contrast, banks with higher credit risk (Mare 2015; Chiorazzo et al. 2018), higher managerial inefficiency (Chiorazzo et al. 2018), higher loan-to-deposit ratios, and higher exposure to non-interest incomes (Stiroh 2004) are more prone to disappear. Overall, the results are not surprising: however, some relationships appear directly related to the period under analysis. In fact, during the period under observation, the squeezing of banks' margins resulting from the outbreak of various crises (sub-prime, sovereign debt, COVID-19), coupled with an unprecedented reduction in interest rates, made the pursuit of efficiency and the containment of credit risks essential. Banks with a high exposure to customer lending and a low ability to raise money from the public savings certainly faced more critical issues during the period analyzed; smaller size proved an additional element of weakness, partly due to the need to be of adequate scale to be able to proceed with effective product and service diversification processes.

The Model 2 column of Table 4 provides the results of estimating the basic model using the microeconomic explanatory variables just described but adding the variable *Contagion*. The underlying idea behind this choice is that the disappearance of small banks in a given geographical area represents the synthesis of critical elements that are manifesting themselves; this is compounded by small banks potentially suffering a stigma effect, that could lead to the disappearance of other intermediaries. As in Poghosyan and Čihak (2011), we find that the existence of more than one case of small bank disappearance in the region where the sample banks are based has an important effect on the market exit of the analyzed banks. Namely, small banks located in regions for which the dummy *Contagion* takes a value equal to 1 have a risk of exit almost 3.97 times higher than that of small banks located in regions without cases of bank exit. The *RAROA* variable remains statistically insignificant in this new specification, while the significance of the remaining bank-specific variables persists.

Model 3 and Model 4 shed light on the contribution of macroeconomic factors in explaining small banks' exits, respectively excluding and including the covariate *Contagion*.

Model 3 shows a negative sign being related to the dynamics of interest rates: hence, an increase in interest rates, consistent with a tighter monetary policy, is associated with a lower risk of small banks disappearing. This result supports the idea that an increase in interest rates raises banks' net interest margin, reducing stress phenomena. The variable related to banking market concentration is not significant in this specification.

In terms of the macroeconomic variables related to the characteristics of the resident population and labor market conditions (proxies for the area's economic health), contributions to the dependent variable are mixed. A higher unemployment rate, an increase in the share of the population being elderly, and higher population density tend to raise the probability of small bank exits occurring. An older population reports less vibrancy in the labor market, which may result in lower output growth and a lesser ability to create wealth. In addition, older customers tend to exhibit more risk-averse characteristics and utilize more basic banking (deposit or investment) instruments.

Adding the variable *Contagion* to this set of macroeconomic covariates (Model 4), *Unemployment* becomes more significant, while the variable *Monetary policy* loses statistical significance. A higher rate of *Unemployment* signals adverse economic conditions in the area; this can reduce the population's capacity for income and wealth accumulation, resulting in a reduced ability to take advantage of banking services (and, for the bank, to raise resources from the area through deposit accounts or offer services with higher value-added

 $^{^2}$ For instance, if we exponentiate the coefficient of the variable *Size* in Model 1, we find a hazard ratio equal to 0.76 which means that the risk of exit for larger banks decreases by 24%. On the contrary, the hazard ratio for the variable *Funding gap* in the same model is equal to 1.049 which implies that the risk of exit increases 5% for larger loans-to-deposit ratio. Therefore, the hazard ratio indicates the change in the risk of disappearance if the variable rises by one unit.

Estimates by type of exit. This table shows in Panel A the complementary log–log model estimations for different types of bank exit obtained by regressing bank disappearance on the micro and macroeconomic lagged variables as described in Table 1. Reported coefficients are not exponentiated. The sample period is 2005–2022. All microeconomic variables are winsorized at the 1% of each tail. Temporal dummy variables are also included in the model. Robust standard errors clustered on the unit of analysis are reported in parentheses. The superscripts ****, ***, ***, and * denote coefficients statistically different from zero at the 0.1%, 1%, 5%, and 10% levels, respectively, in two-tailed tests. This table also displays in Panel B models' goodness of fit concerning the relationship between model predictions and actual events of bank exit. In-sample predictions are computed on the whole sample used to fit the models using a cut-off point equal to 1%, 2%, and 3%; out-of-sample predictions are calculated on a training sample over the period 2005–2017 and a hold-out period covering the years 2018–2022. Panel B also reports the ranking of disappeared banks using the estimated hazard rate, from the least to the most risky. The hazard rate is computed using the estimated coefficients for the banks in-sample over the whole period and out-of-sample. Following Hosmer and Lemeshow (2000) we do not report any R-squared measure (such as the Nagelkerke's pseudo-R-squared) since the authors indicate that a true measure of fit is the one based strictly on a comparison of observed to predicted values from the fitted model.

| Panel A | | |
|----------------------------------|-----------------|----------------|
| | Financial exit | Strategic exit |
| L.Size | -0.054 | -0.394**** |
| | (0.051) | (0.035) |
| L.Capital | -0.045*** | -0.046**** |
| | (0.015) | (0.012) |
| L.Funding gap | 0.070*** | -0.041 |
| | (0.023) | (0.041) |
| L.Credit risk | 0.321**** | 0.189*** |
| | (0.080) | (0.071) |
| L.RAROA | -0.422^{****} | 0.014**** |
| | (0.070) | (0.004) |
| L.Cost to income | 0.007** | 0.023**** |
| | (0.003) | (0.002) |
| L.Income div | 0.009*** | -0.002 |
| | (0.003) | (0.003) |
| L.Monetary policy | -0.402 | -0.178 |
| | (0.286) | (0.189) |
| L.Concentration | 0.0511**** | -0.014 |
| | (0.012) | (0.012) |
| L.Unemployment | 0.016 | -0.004 |
| | (0.012) | (0.013) |
| L.Tertiary education | -0.0007 | -0.001 |
| | (0.008) | (0.007) |
| L.Elderly population | 0.047* | 0.101**** |
| | (0.027) | (0.022) |
| L.Population density | 0.101* | 0.001 |
| | (0.060) | (0.111) |
| Contagion | 1.648**** | 1.682**** |
| | (0.122) | (0.0918) |
| Year dummies | Yes | Yes |
| No. of disappeared banks | 398 | 706 |
| No. of banks | 3,021 | 3,340 |
| Mean hazard of disappeared banks | 0.067 | 0.027 |
| Mean hazard | 0.009 | 0.011 |
| No. of obs | 34,327 | 34,635 |

| Panel B | | |
|--|----------------|----------------|
| | Financial exit | Strategic exit |
| In-sample predictions 1 % cut-off (%) | | |
| Sensitivity | 84.92 | 84.56 |
| Specificity | 76.11 | 53.61 |
| Corr. classified | 76.21 | 54.24 |
| ROC area | 80.52 | 69.09 |
| In-sample predictions 2 % cut-off (%) | | |
| Sensitivity | 73.62 | 74.36 |
| Specificity | 86.35 | 70.96 |
| Corr. classified | 86.21 | 71.03 |
| ROC area | 79.99 | 72.66 |
| In-sample predictions 3 % cut-off (%) | | |
| Sensitivity | 63.57 | 66.86 |
| Specificity | 90.81 | 79.21 |
| Corr. classified | 90.50 | 78.96 |
| ROC area | 77.19 | 73.03 |
| In-sample disappeared banks ranking by decile of the hazard rate (| %) | |
| 1–5 | 4.52 | 13.88 |
| 6 | 2.26 | 4.82 |
| | | |

(continued on next page)

Table 7 (continued)

| Panel B | | |
|--|--|----------------|
| | Financial exit | Strategic exit |
| 7 | 4.77 | 6.94 |
| 8 | 8.04 | 10.48 |
| 9 | 16.08 | 17.99 |
| 10 | 64.32 | 45.89 |
| Out-of-sample predictions 3 % cut-off (hold-out sample 201 | 8–2022) (%) | |
| Sensitivity | 66.89 | 40.83 |
| Specificity | 83.15 | 82.60 |
| Corr. classified | 82.97 | 81.50 |
| ROC area | 75.02 | 61.75 |
| Out-of-sample disappeared banks ranking by decile of the h | azard rate (hold-out sample 2018–2022) (%) | |
| 1–5 | 9.82 | 15.02 |
| 6 | 2.68 | 6.39 |
| 7 | 2.68 | 7.99 |
| 8 | 7.14 | 12.46 |
| 9 | 16.07 | 21.73 |
| 10 | 61.61 | 36.42 |

elements and profitability). Additionally, the variables *Concentration* and *Tertiary education* show statistically significant coefficients, with positive and negative signs respectively. More concentrated national markets appear to lead to an increase in small bank exits at a local level. This outcome (which contradicts the results of Maggiolini and Mistrulli 2005) is in some ways surprising, since in a period of squeezed margins, one might assume that less competitive markets would still allow banks to extract some benefit from dominant positions; however, this role has likely been played by larger banks, which are able to obtain position rents and exploit benefits of scale. From the perspective of supervisory and regulatory policies, this serves as a warning. In fact, over time the authorities have been pushing smaller banks toward aggregation processes, creating the conditions for an increase in the degree of concentration in the financial system; econometric outcomes show that in certain scenarios there are potential risks linked to this strategy, at least as long as players of markedly different sizes compete in the market. The negative sign of *Tertiary education* indicates the presence of individuals with higher educational qualifications, who – given their higher financial literacy – can access more complex services (which are more profitable to the bank) and react more rationally to critical phenomena that may affect the institution (e.g., avoiding bank runs) and ultimately exert a higher degree of discipline on bank management. It is worth noting that the variable *Contagion* is associated with a strongly significant and positive coefficient, suggesting that this covariate provides a noteworthy contribution in explaining small banks' disappearance, even in this novel specification.

The last column of Table 4 (Model 5) shows the results of the analysis of the entire sample of banks using the full set of bank-specific and macroeconomic explanatory variables. Overall, the sign and significance of the coefficients associated with the different covariates do not change from the previously described analyses: only the coefficients associated with *RAROA* and *Unemployment* are not significant from a statistical point of view.

As shown in Table 4, small banks that have disappeared display a higher hazard compared to active ones.

To assess the performance of our models, or in other words, their goodness of fit (as in Fiordelisi and Mare 2013; Betz et al. 2014), we need an evaluation framework that calculates their usefulness in terms of predicting small bank exit. The predictive ability of our models is tested both in- and out-of-sample and reported in Panel B of Table 4. Through these checks, we classify a bank as disappeared if the likelihood of its exit is greater than an optimum cut-off point that we define conservatively and set respectively at equal to 1 %, 2 %, and 3 % for the in-sample predictions. All banks above (or below) that cut-off point are considered as disappeared (or active) banks. Sensitivity quantifies the proportion of disappeared banks that are correctly identified as such. It is worth noting that a higher cut-off point results in a lower number of banks on the list of exited banks, which tends to decrease the Sensitivity of the model or, in other words, increase the so-called Type 1 errors. The latter occur when we classify as active a bank that did not survive.

In our analysis, we also report an indicator of Specificity which measures the ability of the model to correctly classify healthy banks. In this case a higher cut-off point results in a lower number of banks on the list of disappeared banks, which tends to increase the Specificity of the model or, in other words, reduce the so-called Type 2 errors. These occur when we classify as disappeared a bank that did survive or specifically when false alarms of exit are recorded. As policymakers are more concerned about failing to identify bank distress than issuing false alarms, we believe more attention should be given to the model's Sensitivity (e.g., Betz et al. 2014; Chiaramonte et al. 2015).

Finally, we report an indicator proxying the ability of the model to correctly classify disappeared and active banks as a proportion of the total number of banks annually recorded and the area under the receiver operating characteristic (ROC) curve, which measures the overall performance of a binary classifier at the specified cut-off point.

At a 1 % cut-off, all the models, with minor differences, allow us to successfully detect disappeared banks, as demonstrated by the percentage values of the Sensitivity indicator of in-sample tests. However, this comes at the cost of lower Specificity rates and a lesser ability to correctly classify safe and non-surviving banks. Surprisingly, Models 1 and 3 do not exhibit great differences in predictive ability, contrary to Mare (2015) and Betz et al. (2014), thus indicating that the economic and socio-demographic environment in which small banks operate greatly influences the soundness of small banks, independent of their individual financial and economic characteristics. It is worth noting that the inclusion of the variable *Contagion* has a substantial impact on the reduction of Type 2 errors, or in

Estimates for the restricted sample consisting of Austria, Germany, and Italy. This table shows in Panel A the complementary log–log model estimations obtained for a restricted sample consisting of Austria, Germany, and Italy by regressing bank exit on the micro and macroeconomic lagged variables as described in Table 1. Reported coefficients are not exponentiated. The sample period is 2005–2022. All microeconomic variables are winsorized at the 1% of each tail. Temporal dummy variables are also included in the model. Robust standard errors clustered on the unit of analysis are reported in parentheses. The superscripts ****, ***, **, and * denote coefficients statistically different from zero at the 0.1%, 1%, 5%, and 10% levels, respectively, in two-tailed tests. This table also displays in Panel B models' goodness of fit concerning the relationship between model predictions and actual events of bank exit. In-sample predictions are computed on the whole sample used to fit the models using a cut-off point equal to 1%, 2%, and 3%; out-of-sample predictions are calculated on a training sample over the period 2005–2017 and a hold-out period covering the years 2018–2022. Panel B also reports the ranking of disappeared banks using the estimated hazard rate, from the least to the most risky. The hazard rate is computed using the estimated coefficients for the variables in the model using different specifications for the banks in-sample over the whole period and out-of-sample. Following Hosmer and Lemeshow (2000) we do not report any R-squared measure (such as the Nagelkerke's pseudo-R-squared) since the authors indicate that a true measure of fit is the one based strictly on a comparison of observed to predicted values from the fitted model.

| Panel A | | | | | | |
|--|-----------|-----------|-----------|---------|-----------|------------------|
| | Model 1 | Model 2 | Model 3 | Mo | del 4 | Model 5 |
| I Siga | 0 302**** | 0 373**** | | | | 0 /19**** |
| 1.5126 | -0.392 | -0.373 | | | | -0.412 |
| L Comital | (0.030) | (0.031) | | | | (0.033) |
| L.Capitai | -0.091 | -0.080 | | | | -0.08/***** |
| | (0.011) | (0.011) | | | | (0.011) |
| L.Funding gap | 0.027 | 0.014 | | | | -0.003 |
| | (0.024) | (0.027) | | | | (0.037) |
| L.Credit risk | 0.559**** | 0.520**** | | | | 0.434**** |
| | (0.053) | (0.056) | | | | (0.063) |
| L.RAROA | -0.007 | -0.006 | | | | -0.005 |
| | (0.006) | (0.006) | | | | (0.006) |
| L Cost to income | 0.024**** | 0.023**** | | | | 0.024**** |
| Lioust to income | (0.003) | (0.003) | | | | (0.003) |
| I Income dia | 0.007** | (0.003) | | | | 0.005 |
| L.Income alv | 0.007*** | 0.007*** | | | | 0.005 |
| | (0.003) | (0.003) | | | | (0.003) |
| L.Monetary policy | | | -0.542 | -0.0 | 682** | -0.139 |
| | | | (0.344) | (0. | 334) | (0.358) |
| L.Concentration | | | 0.227**** | 0.13 | 35*** | 0.0301 |
| | | | (0.042) | (0.0 |)440) | (0.048) |
| L.Unemployment | | | 0.026*** | 0.03 | 9**** | -0.005 |
| | | | (0.010) | (0. | 011) | (0.013) |
| I. Tertiary education | | | 0.017** | -0 | 002 | -0.021** |
| Life daly calculot | | | (0.008) | (0 | 008) | (0,009) |
| I Eldorby population | | | 0.022* | 0.07 | 4 * * * * | 0.000 |
| L.Elderly population | | | 0.033 | 0.07 | 4 | 0.090 |
| | | | (0.017) | (0. | 020) | (0.021) |
| L.Population density | | | -0.024 | 0. | 015 | 0.129 |
| | | | (0.0646) | (0. | 073) | (0.078) |
| Contagion | | 1.508**** | | 1.66 | 5**** | 1.596**** |
| | | (0.081) | | (0. | 083) | (0.084) |
| Year dummies | Yes | Yes | Yes | Y | les | Yes |
| No. of disappeared banks | 924 | 924 | 924 | g | 24 | 924 |
| No. of banks | 2,805 | 2 805 | 2,805 | 2 | 805 | 2,805 |
| Mean hazard of disappeared banks | _, | _, | _, | _, | | 0.064 |
| Mean hazard | | | | | | 0.023 |
| Ne of obs | 20.074 | 20.074 | 20.074 | 20 | 074 | 0.025 |
| NO. OI ODS | 29,074 | 29,074 | 29,074 | 29 | ,074 | 29,074 |
| Panel B | | | | | | |
| | | Model 1 | Model 2 | Model 3 | Model 4 | Model 5 |
| | | model 1 | model 2 | model o | model | inouci o |
| In-sample predictions 1 % cut-off (%) | | | | | | |
| Sensitivity | | 94.69 | 92.52 | 99.78 | 92.95 | 92.19 |
| Specificity | | 16.29 | 35.05 | 1.15 | 25.59 | 36.78 |
| Corr. classified | | 18.78 | 36.87 | 4.28 | 27.73 | 38.54 |
| BOC area | | 55 49 | 63.78 | 50.47 | 59.27 | 64 48 |
| In-sample predictions 2 % cut-off (%) | | 00115 | 001/0 | 00117 | 03127 | 01110 |
| Consistential | | 92.05 | 00.75 | 04 71 | 77 66 | 00.10 |
| Sensitivity | | 83.95 | 82.75 | 84.71 | //.00 | 82.10 |
| Specificity | | 45.49 | 60.92 | 31.94 | 61.53 | 60.86 |
| Corr. classified | | 46.70 | 61.62 | 33.61 | 62.04 | 61.53 |
| ROC area | | 64.72 | 71.84 | 58.32 | 69.59 | 71.48 |
| In-sample predictions 3 % cut-off (%) | | | | | | |
| Sensitivity | | 68.76 | 75.16 | 63.12 | 75.49 | 75.05 |
| Specificity | | 66.18 | 69.97 | 59.51 | 65.31 | 70.76 |
| Corr. classified | | 66.26 | 70.13 | 59.63 | 65.63 | 70.89 |
| ROC area | | 67 47 | 72.56 | 61.32 | 70.40 | 72.91 |
| In-sample disappeared banks ranking by | | 07.17 | , 2.00 | 01.02 | , 5.10 | / 21. / 1 |
| decile of the bagand note (04) | | | | | | |
| | | 24.70 | 10.46 | 07.14 | 16.04 | 12.02 |
| 1–0 | | 24.79 | 13.40 | 2/.14 | 16.94 | 13.02 |
| | | | | | (continue | ed on next page) |

Table 8 (continued)

| Panel B | | | | | |
|---|---------|---------|---------|---------|---------|
| | Model 1 | Model 2 | Model 3 | Model 4 | Model 5 |
| 6 | 6.20 | 4.32 | 9.53 | 4.52 | 5.31 |
| 7 | 8.97 | 6.16 | 10.20 | 8.66 | 6.94 |
| 8 | 9.94 | 9.68 | 14.15 | 16.65 | 11.39 |
| 9 | 14.32 | 15.54 | 17.23 | 20.69 | 17.90 |
| 10 | 35.79 | 35.84 | 21.75 | 32.53 | 45.44 |
| Out-of-sample predictions 3 % cut-off (hold-out sample 2018–2022) (%) | | | | | |
| Sensitivity | 37.71 | 55.87 | 21.79 | 76.82 | 63.41 |
| Specificity | 86.83 | 79.74 | 85.90 | 59.09 | 72.94 |
| Corr. classified | 84.99 | 78.85 | 83.51 | 59.75 | 72.58 |
| ROC area | 62.27 | 67.80 | 53.85 | 67.95 | 68.17 |
| Out-of-sample disappeared banks ranking by decile | | | | | |
| of the hazard rate (hold-out sample 2018–2022) (%) | | | | | |
| 1–5 | 31.52 | 16.06 | 47.89 | 18.67 | 16.61 |
| 6 | 4.55 | 5.45 | 7.53 | 7.23 | 6.90 |
| 7 | 10.30 | 9.70 | 7.83 | 14.46 | 9.40 |
| 8 | 9.09 | 14.24 | 9.94 | 15.96 | 12.85 |
| 9 | 12.42 | 16.67 | 13.55 | 18.07 | 14.73 |
| 10 | 32.12 | 37.88 | 13.25 | 25.60 | 39.50 |
| | | | | | |

other words the issue of false alarms on the soundness of small banks, which in turn also increases the percentage of banks correctly classified by the models. By including all the selected explanatory variables in Model 5, its overall predictive ability improves, although with a lower Sensitivity rate. If the cut-off is raised to 3 %, as expected, the Sensitivity decreases but the overall predictive capacity of the different models improves, so that false alarms and the supervisory authorities' consequent control work diminish. In comparison with the results of Männasoo and Mayes (2009), Fiordelisi and Mare (2013), Betz et al. (2014), and Mare (2015) the in-sample predictive capacity of our models (at 3 % cut-off) is slightly lower than that reported by the aforementioned authors. For example, their Sensitivity rate is between 70 % and 90 %, as is the percentage of cases correctly classified by their models. However, these authors use very different samples from ours. Therefore, we believe that comparability is limited, given that ours is, as far as we know, a unique study both in terms of the sample used and the models' specifications.

As a further measure of goodness of fit, we report the in-sample ranking of the banks using the estimated probability of exit or the hazard rate. The hazard rate is computed using the estimated coefficients for the variables in the different model specifications. The ranking shows that most of the banks that left the market are in the highest deciles, which is not surprising given the average value of the risk of the small banks that disappeared. Moreover, in Panel B we report two sets of diagnostics for the out-of-sample predictions. In the first instance, we used a training sample that covers the period 2005-2017 while the hold-out sample refers to the years 2018–2022. We believe that this last sample could be valuable for evaluating the predictive capabilities of our models, given that it includes the effects of the 2019 pandemic crisis. We instead opted for a large training sample from 2005 to 2017 which covers multiple episodes of financial crisis to which small European banks were exposed. Further, we employ a training sample consisting of banks located in Austria, Germany, and Italy and a hold-out sample consisting of banks in the remaining countries. This choice is motivated by the relevance and variety that small banks assume in the three countries mentioned, which should provide a valuable assessment of how well the models might do if applied to new data. We rely on the sole 3 % cut-off for the sake of brevity in both out-of-sample predictions and show that the overall predictive power of Model 5 is quite satisfactory except in terms of Sensitivity. Hazard rankings are also presented regarding the in-sample and out-of-sample forecasts in Panel B of Table 4.

A final element to highlight concerns the incremental effect that the Contagion variable has on non-surviving banks' probability of exiting the market, compared to estimates in which this variable is not present. Its inclusion helps to better classify healthy banks which remain sound even in the presence of non-surviving banks in the region in which they are located.

3.1.1. Digging into banking size: the role of micro and macro determinants of bank survival

As stated earlier, our analysis focuses on small banks (with less than 5 billion euros in total assets); although the size threshold is very low, there is nevertheless a great deal of variability in terms of operating strategies beneath it. Table 5, for this purpose, provides separate estimates for three sub-samples with intermediate trigger values of 250 million and 1 billion euros.

The first column provides the coefficients associated with the bank-specific and macroeconomic variables for banks with total assets of less than 250 million euros. It emerges that the probability of exit is higher for small, less capitalized, more inefficient banks, that are more exposed to credit risks and show higher loan-to-deposit ratios. This confirms what was stated in the previous comments, where the elements of greater vulnerability related to the period under analysis were provided. The macroeconomic environment also plays a significant role: the probability of the occurrence of the dependent variable "Event" increases as the population with advanced educational degrees shrinks, as the population ages, and as monetary policy is looser. The coefficient associated with the variable Contagion is positive and strongly significant.

Columns 2 and 3 of Table 5, which are devoted respectively to estimates for the sub-sample of banks with total assets between 250 million and 1 billion euros, and banks with total assets between 1 and 5 billion euros, show coefficients that are only partially in line with those described previously. As regards the microeconomic explanatory factors of exit, the coefficient associated with the size of the bank loses statistical significance for the sub-sample of bigger banks. Therefore, the variable Size emerges as a significant factor in explaining

Estimates by sub-periods. This table shows in Panel A the complementary log–log model estimations obtained for the two sub-periods 2005–2012 and 2013–2022 by regressing bank exit on the micro and macroeconomic lagged variables as described in Table 1. Reported coefficients are not exponentiated. All microeconomic variables are winsorized at the 1% of each tail. Temporal dummy variables are also included in the model. Robust standard errors clustered on the unit of analysis are reported in parentheses. The superscripts ****, ***, ***, and * denote coefficients statistically different from zero at the 0.1%, 1%, 5%, and 10% levels, respectively, in two-tailed tests. This table also displays in Panel B models' goodness of fit concerning the relationship between model predictions and actual events of bank exit. In-sample predictions are computed on the whole sample used to fit the models using a cut-off point equal to 1%, 2%, and 3%. Panel B also reports the ranking of disappeared banks using the estimated hazard rate, from the least to the most risky. The hazard rate is computed using the estimated coefficients for the variables in the model using different specifications for the banks in-sample over the two sub-periods. Following Hosmer and Lemeshow (2000) we do not report any R-squared measure (such as the Nagelkerke's pseudo-R-squared) since the authors indicate that a true measure of fit is the one based strictly on a comparison of observed to predicted values from the fitted model.

| Panel A | | |
|----------------------------------|-------------------|-------------------|
| | From 2005 to 2012 | From 2013 to 2022 |
| L.Size | -0.226**** | -0.317**** |
| | (0.064) | (0.033) |
| L.Capital | -0.072*** | -0.055**** |
| | (0.026) | (0.011) |
| L.Funding gap | 0.015 | 0.049* |
| | (0.039) | (0.027) |
| L.Credit risk | 0.474**** | 0.415**** |
| | (0.123) | (0.057) |
| L.RAROA | 0.002 | -0.010* |
| | (0.011) | (0.006) |
| L.Cost to income | 0.023**** | 0.020**** |
| | (0.005) | (0.002) |
| L.Income div | 0.002 | 0.006** |
| | (0.006) | (0.002) |
| L.Monetary policy | -0.553 | -0.127 |
| | (0.360) | (0.176) |
| L.Concentration | 0.083**** | 0.013 |
| | (0.018) | (0.010) |
| L.Unemployment | 0.032* | 0.000 |
| | (0.017) | (0.010) |
| L.Tertiary education | -0.006 | -0.005 |
| | (0.013) | (0.007) |
| L.Elderly population | 0.116*** | 0.053*** |
| | (0.042) | (0.019) |
| L.Population density | -0.031 | 0.104* |
| | (0.142) | (0.062) |
| Contagion | 1.722**** | 1.625**** |
| | (0.153) | (0.083) |
| Year dummies | Yes | Yes |
| No. of disappeared banks | 223 | 881 |
| No. of banks | 2,556 | 3,501 |
| Mean hazard of disappeared banks | 0.057 | 0.102 |
| Mean hazard | 0.020 | 0.039 |
| No. of obs. | 11,411 | 23,622 |
| Panel B | | |

| | From 2005 to 2012 | From 2013 to 2022 |
|---|-------------------|--------------------------|
| In-sample predictions 1 % cut-off (%) | | |
| Sensitivity | 85.65 | 92.87 |
| Specificity | 49.69 | 27.57 |
| Corr. classified | 50.39 | 30.10 |
| ROC area | 67.67 | 60.22 |
| In-sample predictions 2 % cut-off (%) | | |
| Sensitivity | 62.78 | 85.01 |
| Specificity | 76.60 | 52.62 |
| Corr. classified | 76.33 | 53.88 |
| ROC area | 69.69 | 68.81 |
| In-sample predictions 3 % cut-off (%) | | |
| Sensitivity | 57.40 | 79.69 |
| Specificity | 82.17 | 61.95 |
| Corr. classified | 81.68 | 62.63 |
| ROC area | 69.78 | 70.82 |
| In-sample disappeared banks ranking by decile of th | e hazard rate (%) | |
| 1–5 | 14.80 | 14.51 |
| 6 | 8.07 | 5.44 |
| 7 | 8.52 | 7.86 |
| 8 | 9.87 | 11.97 |
| | | (continued on next page) |

Table 9 (continued)

| Panel B | From 2005 to 2012 | From 2013 to 2022 |
|---------|-------------------|-------------------|
| 9 | 16.14 | 16.32 |
| 10 | 42.60 | 43.89 |

exit for smaller banks: dwarf banks, unable to exploit the beneficial effect that stems from economies of scale, are most exposed to the risk of disappearance. The level of *Capital* maintains its statistical significance only for banks with total assets below the threshold of 1 billion euros; the same occurs for *Credit risk*, while efficiency remains a significant covariate in all the estimations. Conversely, the *Funding gap* and risk-adjusted profitability gain statistical significance for the larger banks sub-sample. Overall, these outcomes can be explained if one considers that these two sub-samples of larger (but still small) banks have easier and more frequent access to financial markets for funding; this leads to a different set of signaling indicators that can explain the dynamics in their disappearance.

As regards macroeconomic covariates, we observe a scattered pattern of statistically significant coefficients. Only *Monetary policy* and *Contagion* show constant behavior, being associated with strongly significant coefficients. The latter covariate deserves a further note: the magnitude of the coefficients shown in Table 5 increases for smaller size sub-samples, indicating that the disappearance of banks in a specific area generates spillovers that exert a higher pressure on smaller banks.

The goodness of fit analyses contained in Panel B of Table 5 show how the overall predictive quality of the model soars as the size threshold of the sub-sample of analyzed banks increases. This highlights how the disappearance of extremely small banks is a poorly predictable event, which indirectly justifies the pressure from regulators and supervisors for M&A phenomena to increase the size threshold of smaller banks. The out-of-sample goodness of fit is generally in line with that reported in Table 4.

3.1.2. Do the business models of small banks all perform equally?

The sample analyzed is made up of many commercial, savings, and cooperative banks; these three categories reflect differences between business models, from which it is reasonable to hypothesize that different competitive behaviors (and therefore different levels of financial vulnerability) may arise. Fiordelisi and Molyneux (2006) state that commercial banks are entities that are predominantly oriented towards creating value for their shareholders (shareholder value-oriented banks); savings, and to an even greater extent cooperative banks, do not focus on maximizing profits, but pursue objectives of the welfare of different stakeholders at a local level (stakeholder value-oriented banks). Among the latter, there are certainly the bank's shareholders (often not surprisingly defined as "members"), but also associations and other local structures often linked to the non-profit sector (Fiordelisi et al. 2022). These characteristics make cooperative banks more inclined to prioritize relationship banking policies, reducing agency costs (Rasmusen 1988), and maintaining a low level of risk and a high level of capitalization. Their small size, however, leads to higher levels of income concentration and makes them less able to attract high-profile managerial figures (Fiordelisi et al. 2022). It is, therefore, reasonable to expect that the type of business model leads to differences in the financial results of the analyzed banks; however, given the advantages and disadvantages linked to the choice of a shareholder value- or stakeholder value-oriented model, the final result in terms of disappearance rates must be empirically determined.

Table 6 shows estimates for sub-samples of banks that are homogeneous from a business model perspective. The average risk of commercial banks and disappeared banks that fall into this sub-sample is the highest among the different business models.

Commercial banks (first column) show significant coefficients associated with three bank-specific variables, namely *Credit risk*, *RAROA* and *Cost to income*. Among the macroeconomic variables, market concentration, unemployment, and the dummy signaling the presence of multiple non-surviving banks show significant coefficients.

As regards cooperative and savings banks, the outcome of the analysis provides a consistent picture for the two sub-samples. In particular, the survival of banks appears to be facilitated by greater size, higher levels of capitalization, lower credit risk, and greater efficiency. The proxy for income diversification is associated with a statistically significant coefficient only for the cooperative banks sub-sample. Among the macroeconomic variables, statistically significant coefficients are found for *Elderly population* and *Population density*, while *Monetary policy* is associated with a statistically significant coefficient only for cooperative banks. However, in all estimates reported in Table 6, the *Contagion* variable exhibits a very high statistical significance.

Overall, the predictive ability of the model appears higher for savings banks but at the cost of a lower Sensitivity rate.

3.1.3. Financial versus strategic exits: micro and macro determinants at work

The concepts of survival or market exit can be easily represented through binary variables from Stigler's (1958) perspective. However, Chiorazzo et al. (2018) recall that this approach, while extremely simple to apply in empirical research, neglects the fact that the disappearance of an operator from the market can occur either for financial reasons (for example episodes of failure) or for strategic reasons (for example through M&A operations carried out with other banks, not with the aim of rescue, but for strategic purposes, including preventing future situations of distress). This insight has practical implications which are not negligible: considering financial and strategic exit as identical events paradoxically leads to the risk of confusing entities that have exited the market due to serious financial distress and entities that have become the object of other companies' interest due to their virtues (including in terms of profitability).

As stated previously, to separate financial and strategic exits we rely on the annual Z-score of the banks; more precisely, banks that show a Z-score in the lower quartile of the distribution of this variable are associated with financial exit, while the other banks are associated with strategic exit. 31.6 % of the banks included in our sample occupy the lower quartile of the Z-score; among them, 76.5 % have been associated with a financial exit. Conversely, 25.7 % of the banks included in the sample are found in the top quartile; among

these, 18.6 % are associated with financial exit.

Following this approach, Table 7 provides separate estimates for sub-samples attributable to financial and strategic exit.

The first column, devoted to financial exits, shows that the probability of this outcome rises for banks that are less capitalized, less efficient, less profitable, more exposed to non-interest incomes, and that have a higher funding gap and credit risk. Among the macroeconomic covariates, higher market concentration (Chiorazzo et al. 2018), population age, and density increase the probability of a financial exit. Again, the coefficient associated with the variable *Contagion* is strongly significant.

The second column, related to strategic exits, shows some peculiarities; in this specification, *Size* is associated with a negative and significant coefficient while we observe a positive coefficient being associated with *RAROA*. The latter result seems to imply that small banks with increasing risk-adjusted profitability are more attractive for strategic exits, such as mergers or acquisitions. Moreover, the proxies for funding gap and income diversification do not show a significant impact on this kind of market exit. Among the macro-economic covariates, only *Elderly population* and *Contagion* are associated with (strongly) significant coefficients.

Diagnostics of the predictive power in-sample provide slightly better results for cases of financial exit.

3.2. Robustness checks

Within the area under examination, some countries (namely, Austria, Germany, and Italy) are characterized by a high number (and significant market share) of small banks: the estimation models described above were therefore applied to a sub-sample of banks belonging to these three countries to test whether the results of the econometric analysis remain unchanged and are not driven by other countries. Table 8 provides the outcome of this additional test.

Overall, the coefficients associated with the different explanatory variables retain their sign when compared with those in Table 4; there are, however, slight variations in the statistical significance found. Among the bank-specific variables, the explanatory contribution of the *Funding gap* is lost in all the specifications; the same occurs with the coefficient associated with *Population density* among the macroeconomic variables.

As reported in Panel B of Table 8, the results of the tests relating to the predictive ability of our in-sample models do not change significantly compared to what was found for the entire sample. Out-of-sample predictions are not reported given that the sample of banks in Austria, Germany, and Italy is used as a training sample to evaluate its predictive capacity on the hold-out sample represented by the remaining European countries (see Table 4).

In a further test, we consider the sub-sample of banks excluding Germany; the main results proved unchanged, especially for microeconomic determinants and the variable *Contagion*. These outcomes are available on request.

Our work covers a particularly lengthy period, during which the economic and financial conditions in which banks operate have changed significantly. In this regard, the topic of the level of interest rates (and more generally of looser monetary policy) is particularly evocative. In fact, in 2012, Mario Draghi's famous speech (remembered for the renowned quote "Whatever it takes") gave birth to a more active approach by the European Central Bank, leaving room for non-conventional monetary policies; moreover, since 2014 many countries have experienced particularly low or even negative interest rates policy (NIRP). This has resulted in a widespread contraction of interest income margins for the banking sector (Borio and Gambacorta 2017; Claessens et al. 2018; Committee on the Global Financial System 2018). In some cases, however, the ability to diversify revenue sources has allowed banks to offset losses in interest income with non-interest income flows (Altavilla et al. 2018; Lopez et al. 2020). Considering the limited ability of small banks to diversify their revenue sources, as they focus mainly on deposit-taking and lending, it is conceivable that the pattern of bank exits was different before and during the implementation of non-conventional monetary policies. Hence, Table 9 divides the time span under scrutiny into two sub-periods: the first between 2005 and 2012 and the second from 2013 to 2022.

Overall, the microeconomic covariates are associated with coefficients having the same sign as those shown in Table 4 (Model 5), although the significance sometimes differs. The banks that exited the market during both sub-periods were smaller, less capitalized, less efficient, and more exposed to credit risks. The proxy for income diversification is associated with a significant and positive coefficient in the period 2013–2022; this could indicate that during a period of squeezing margins small banks tried to diversify their revenues, but the overall outcome was a higher probability of exiting the market. In this same sub-period, banks with lower profitability and higher funding gaps have been associated with higher exit rates.

Among the macroeconomic variables, it is shown that *Elderly population* and *Contagion* play a significant role in both the subperiods. *Concentration* is associated with a positive and significant coefficient only during the period 2005–2012, suggesting that in those years more concentrated markets increased the probability of experiencing a strategic or financial exit. A weak and positive significant coefficient associated with *Unemployment* is also found during the first sub-period; the same occurs for *Population density* for the years 2013–2022. The mean hazards for this latter period are much higher than those recorded in the previous period.

Diagnostics to ascertain the predictive power of the models provide better in-sample results for the 2005–2012 period, when – despite the outbreak of the international financial crisis – the macroeconomic and financial environment nevertheless presented more typical conditions than in the following decade.

4. Policy implications

Small banks play a fundamental role in European countries; the exit of these intermediaries therefore represents a phenomenon that must be managed carefully. The focus placed by regulatory and supervisory bodies on significant entities has often led to an underestimation of the effects that can derive from the disappearance of small banks. These latter events can have systemic outcomes that are far greater than expected, with negative repercussions for the areas in which small banks operate and for the financial system as a whole. This opinion is shared by the European Commission's recent Impact Assessment Report (European Commission 2023), according to which the distress of even small banks may lead to the destruction of economic value locally/regionally or the disruption of depositors' confidence, especially for small geographical areas. For these reasons, it is extremely important to have predictive tools that allow us to appreciate the exposure of small banks to idiosyncratic risks and/or those that are attributable to the local/regional economic context of these banks.

The predictive capacity of the models run in this paper to identify small banks' exits highlights very low levels of Type 1 errors, while Type 2 errors are much more frequent: this means that selected variables can intercept almost all the cases that have proved to be truly problematic but also generate many false signals concerning banks that can recover from temporary difficulties. Although from the point of view of supervisory authorities, the minimization of Type 1 errors is of greater importance than Type 2 ones, nevertheless our evidence indicates how complex it is to develop models that optimize both error levels, at least where small banks are concerned. This increased workload to which the authorities are exposed can also explain why bank supervisors push toward a process of aggregation, including through the formation of highly integrated groups among small banks. In this case, however, the parent company should act not as a mere manager of geographically diversified assets (small local banks), but should rather act for crisis prevention or resolution purposes, considering the macroeconomic effects that may arise at a territorial level, as also noted by the above-mentioned European Commission's report.

As demonstrated by our main findings, the possible occurrence of situations in which local banks disappear confirms the importance of the role of the capital cushion held and the level of operational efficiency on which small banks can rely. This is an important outcome both for its managerial implications (a competitor's crisis tends to undermine a bank's soundness regardless of its financials) and for its supervisory ones (mapping information at the territorial level takes on greater significance in predicting disappearance than information related only to analyzing bank balance sheets).

From a regulatory point of view, our results confirm that the basic approach of the Basel III framework, focused on strengthening bank capital, is correct. The existence of higher levels of capitalization reduces the probability of exit from the market of the banks in our sample, both for financial and for more strategically related reasons. However, the risk-based capital requirements related to the new calculation of Risk Weighted Assets (RWA) for credit and market risks envisaged by the finalization of Basel III require further organizational and management efforts, in terms of risk management, compliance and, once again, more capital. This could significantly increase regulatory costs (Gržeta et al. 2023), with a negative impact on the level of banks' cost to income ratio: in light of our empirical results, this effect raises potential concerns regarding the survival of small banks.

The finalization of Basel III seems to make it complex for small banks to choose stand-alone solutions. Appropriate aggregation phenomena, often hoped for by the authorities (e.g. Enria 2023; Fernandez-Bollo et al. 2021), especially where pockets of inefficiency tend to be created through phenomena of overbanking capacities (Gardò and Klaus 2019), can be functional in achieving economies of scale linked to costs of compliance as well as gaining easier access to capital markets. Recent studies, although limited to national experiences (in this specific case, Italy) show how aggregation into integrated groups (Cornée et al. 2018) can allow small banks to recover in terms of efficiency (Beccalli et al. 2023), gain protection regarding the stability of the banks themselves (Single Resolution Board 2023), while safeguarding the traditional biodiversity of the banking system.

5. Conclusions

Based on a wide sample of banks headquartered in 27 European countries over the period 2005–2022, this paper tests the influence that microeconomic and macroeconomic variables have on small banks' probability of exit from the market, evaluating the predictive power of the explanatory models employed.

One of the first contributions of our study relates to the fact that microeconomic and macroeconomic factors have a similar signaling value in identifying potential exits for small banks; in a certain sense, the territorial link seems to blur the boundaries of a bank, leading it to absorb the virtues and critical issues of the territory in which it is located. Models incorporating microeconomic and macroeconomic covariates are good descriptors of the pattern of events observed in our database: this means that both the micro and the macroeconomic data are predictive of the exit of small banks.

A further contribution of our work concerns the importance of the soundness of peer banks in local areas, which appears to be crucial to the survival of small banks. We show that the occurrence of contagion episodes exposes small banks to a higher risk of exit and has a significant predictive value that also minimizes exposure to Type 2 errors. The relationship of trust that exists with customers and the apparent greater fragility that the public could associate with dwarf banks exposes them to critical factors that are at least partially exogenous to corporate strategies. Since our work typically has a local focus, the contagion effects deriving from the concomitant crisis of national champions (or small banks in nearby regions) have not been considered here: this constitutes a further stimulus for digging deeper into the channels of contagion transmission.

Our work also highlights that the role played by the direction of monetary policy is not always significant across all different specifications and sub-samples. This finding is, for instance, not in line with the recent scientific debate and empirical evidence in this regard, which have ascribed the greater fragility to which small banks are exposed to prolonged periods of low or even negative rates.

This study opens the door to subsequent research on the effect that belonging to a differently integrated network or group can have on the survival of small banks. The researches available so far are territorially limited (usually to a single country) and focused on specific explained variables such as cost economies or profitability.

Additionally, the available empirical studies (including our work) do not account for the potential crowding out effect – or the beneficial impact on cost efficiency – played by technology on small independent banks, due to the inconsistent availability of reliable data on this topic at a local level. Finally, given the specific features of the sample employed for this study – i.e. a unique dataset of multi-country small banks– our results cannot be fully compared with those of existing literature.

CRediT authorship contribution statement

Federica Poli: Writing – original draft, Supervision, Methodology, Data curation. **Simone Rossi:** Writing – original draft, Methodology, Data curation, Writing – review & editing. **Mariarosa Borroni:** Writing – original draft, Supervision, Data curation.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Data availability

Data will be made available on request.

Acknowledgments

The authors thank Jonathan A. Batten (the editor), as well as two anonymous reviewers for their constructive and insightful comments. We are also grateful for the helpful suggestions received from participants at the Conference "The Future of Financial Mutuals" at Università Cattolica del Sacro Cuore (December 1st., 2023, Milan).

References

Alessandri, P., Nelson, B.D., 2015. Simple Banking: Profitability and the Yield Curve. Journal of Money, Credit and Banking 47 (1), 143–175. https://doi.org/ 10.1111/jmcb.12172.

- Altavilla, C., Burlon, L., Giannetti, M., Holton, S., 2022. Is there a zero lower bound? The effects of negative policy rates on banks and firms. Journal of Financial Economics 144 (3), 885–907. https://doi.org/10.1016/j.jfineco.2021.06.032.
- Altavilla, C., Boucinha, M., Peydro, J.L., 2018. Monetary policy and bank profitability in a low interest rate environment. Econ. Policy 33 (96), 531–586. Available at https://econpapers.repec.org/RePEc:ecb:ecbwps:20172105 (accessed 26 July 2024).
- Angeloni, I., Faia, E., Lo Duca, M., 2015. Monetary policy and risk taking. Journal of Economic Dynamics and Control 52, 285–307. https://doi.org/10.1016/j. jedc.2014.12.001.
- Arena, M., 2008. Bank failures and bank fundamentals: A comparative analysis of Latin America and East Asia during the nineties using bank-level data. Journal of Banking & Finance 32, 299–310. https://doi.org/10.1016/j.jbankfin.2007.03.011.
- Ayadi R., Arbak E., Schmidt R.H., Carbò Verde S., Fernandez F.R., 2009. Investigating Diversity in the Banking Sector in Europe. CEPS-- Centre for European Policy Studies, Brussels. Available at https://www.ceps.eu/ceps-publications/investigating-diversity-banking-sector-europe-performance-and-role-savings-banks/ (accessed 26 July 2024).
- Ayadi, R., Llewellyn, D. T., Schmidt, R. H., Arbak, E., De Groen, P.W., 2010. Investigating diversity in the banking sector in Europe: Key developments, performance and role of cooperative banks. Centre for European Policy Studies, Brussels. Available at https://www.ceps.eu/ceps-publications/investigating-diversity-bankingsector-europe-key-developments-performance-and-role/ (accessed 26 July 2024).
- Barra, C., Ruggiero, N., 2021. Do microeconomic and macroeconomic factors influence Italian bank credit risk in different local markets? Evidence from cooperative and non-cooperative banks. Journal of Economics and Business 114, 105976. https://doi.org/10.1016/j.jeconbus.2020.105976.
- Barth, J.R., Cebula, R.J., Xu, J., 2023. Do minority banks perform better or worse than non-minority banks? Applied Economics 1–17. https://doi.org/10.1080/00036846.2023.2167922.
- Beccalli, E., Frantz, P., 2013. The determinants of mergers and acquisitions in banking. Journal of Financial Services Research 43, 265–291. https://doi.org/10.1007/s10693-012-0138-y.
- Beccalli, E., Rossi, L., Viola, A., 2023. Network vs integrated organizational structure of cooperative banks: Evidence on the Italian reform. International Review of Financial Analysis 89 (2023), 102812. https://doi.org/10.1016/j.irfa.2023.102812.
- Beck, T., De Jonghe, O., Schepens, G., 2013. Bank competition and stability: Cross-country heterogeneity. Journal of Financial Intermediation 22, 218–244. https://doi.org/10.1016/j.jfi.2012.07.001.
- Bellucci, A., Borisov, A., Zazzaro, A., 2013. Do banks price discriminate spatially? Evidence from small business lending in local credit markets. Journal of Banking & Finance 37, 4183–4197. https://doi.org/10.1016/j.jbankfin.2013.06.009.
- Berger, A.N., Bouwman, C.H.S., 2013. How does capital affect bank performance during financial crises? Journal of Financial Economics 109, 146–176. https://doi. org/10.1016/j.jfineco.2013.02.008.
- Betz, F., Oprică, S., Peltonen, T.A., Sarlin, P., 2014. Predicting distress in European banks. Journal of Banking & Finance 45, 225–241. https://doi.org/10.1016/j. jbankfin.2013.11.041.
- Borio, C., Gambacorta, L., 2017. Monetary policy and bank lending in a low interest rate environment: diminishing effectiveness? Journal of Macroeconomics. 54 (B), 217–231. https://doi.org/10.1016/j.jmacro.2017.02.005.
- Borio, C., Zhu, H., 2012. Capital regulation, risk-taking and monetary policy: A missing link in the transmission mechanism? Journal of Financial Stability 8, 236–251. https://doi.org/10.1016/j.jfs.2011.12.003.
- Boyd, J.H., De Nicolò, G., 2005. The Theory of Bank Risk Taking and Competition Revisited. The Journal of Finance 60, 1329–1343. https://doi.org/10.1111/j.1540-6261.2005.00763.x.
- Boyd, J.H., Graham, S.L., 1986. Risk, Regulation, and Bank Holding Company Expansion Into Nonbanking. QR 10. https://doi.org/10.21034/qr.1021.
- Brissimis, S. N., Delis, M. D. and Iosifidi, M. 2014. Bank market power and monetary policy transmission, International Journal of Central Banking, vol. 10, no. 4, 173–213. Available at https://www.ijcb.org/journal/ijcb14q4a6.pdf (accessed 26 July 2024).
- Calice, G., Gam, Y.K., 2023. US National Banks and Local Economic Fragility. J Financ Serv Res 63, 313–338. https://doi.org/10.1007/s10693-022-00382-3.
- Caselli, G., Figueira, C., Nellis, J.G., 2020. Ownership diversity and the risk-taking channel of monetary policy transmission. Cambridge Journal of Economics 44, 1329–1364. https://doi.org/10.1093/cje/beaa011.
- Chen, M.H., Dey, D.K., Shao, Q.M., 1999. A new skewed link model for dichotomous quantal response data. Journal of the American Statistical Association 94 (448), 1172–1186. https://doi.org/10.1080/01621459.1999.10473872.

Chernykh, L., 2014. Dwarf banks. Journal of Banking & Finance 38, 31–40. https://doi.org/10.1016/j.jbankfin.2013.09.014. https://doi.org/10.1016. https://doi.0016. https://doi.0016. https://doi.0016. https://doi.0016. https://doi.0016. https://doi.0016

- Chiaramonte, L., Croci, E., Poli, F., 2015. Should we trust the Z-score? Evidence from the European Banking Industry. Global Finance Journal 28, 111–131. https://doi.org/10.1016/j.gfj.2015.02.002.
- Chiorazzo, V., D'Apice, V., DeYoung, R., Morelli, P., 2018. Is the traditional banking model a survivor? Journal of Banking & Finance 97, 238–256. https://doi.org/ 10.1016/j.jbankfin.2018.10.008.

- Claessens, S., Coleman, N., Donnelly, M., 2018. 'Low-For-Long' interest rates and banks' interest margins and profitability: Cross-country evidence. Journal of Financial Intermediation 35, 1–16. https://doi.org/10.1016/j.jfi.2017.05.004.
- Clark, E., Mare, D.S., Radić, N., 2018. Cooperative banks: What do we know about competition and risk preferences? Journal of International Financial Markets, Institutions and Money 52, 90–101. https://doi.org/10.1016/j.intfin.2017.09.008.
- Cleary, S., Hebb, G., 2016. An efficient and functional model for predicting bank distress: In and out of sample evidence. Journal of Banking & Finance 64, 101–111. https://doi.org/10.1016/j.jbankfin.2015.12.001.
- Coccorese, P., Shaffer, S., 2021. Cooperative banks and local economic growth. Regional Studies 55, 307–321. https://doi.org/10.1080/00343404.2020.1802003. Cole, R.A., White, L.J., 2012. Déjà vu all over again: The causes of US commercial bank failures this time around. Journal of Financial Services Research 42, 5–29.
- https://doi.org/10.1007/s10693-011-0116-9.
- Coleman, J.S., 1988. Social Capital in the Creation of Human Capital. American Journal of Sociology, 94, 95–120. Available at http://www.jstor.org/stable/2780243 (accessed 26 July 2024).
- Committee on the Global Financial System, 2018. Financial stability implications of a prolonged period of low interest rates. CGFS papers, 61. Available at https://www.bis.org/publ/cgfs61.pdf (accessed 26 July 2024).
- Contreras, S., Ghosh, A., Hasan, I., 2023. The effect of bank failures on small business loans and income inequality. Journal of Banking & Finance 146, 106690. https://doi.org/10.1016/j.jbankfin.2022.106690.
- Cornée, S., Fattobene, L., Migliorelli, M., 2018. An Overview of Cooperative Banking in Europe. In: Migliorelli, M. (Ed.), New Cooperative Banking in Europe. Strategies for Adapting the Business Model Post Crisis. Palgrave MacMillan, Cham. https://doi.org/10.1007/978-3-319-93578-2.
- Cox, D.R., 1972. Regression Models and Life Tables. Journal of the Royal Statistical Society Series B 34, 187–220. https://doi.org/10.1111/j.2517-6161.1972.tb00899.
- Cox, R.A.K., Kimmel, R.K., Wang, G.W.Y., 2017. Proportional Hazards Model of Bank Failure: Evidence from USA. Journal of Economic & Financial Studies Vol. 05, No. 03: June, 35–45. https://doi.org/10.18533/jefs.v5i03.290.
- Czado, C., Santner, T.J., 1992. The effect of link misspecification on binary regression inference. Journal of Statistical Planning and Inference 33, 213–231. https://doi.org/10.1016/0378-3758(92)90069-5.
- De Nicolò, G., Lucchetta, M., 2009. Financial Intermediation, Competition and Risk: A General Equilibrium Exposition. IMF Working Paper WP/09/105, May. Available at https://www.imf.org/en/Publications/WP/Issues/2016/12/31/Financial-Intermediation-Competition-and-Risk-A-General-Equilibrium-Exposition-22931 (accessed 26. July 2024).
- Demertzis, M., Wolff, G.B., 2016. What impact does the ECB's quantitative easing policy have on bank profitability? Bruegel Policy Contribution, No. 2016/20. Available at https://www.bruegel.org/policy-brief/what-impact-does-ecbs-quantitative-easing-policy-have-bank-profitability (accessed 26 July 2024).
- Demsetz, R.S., Strahan, P.E., 1997. Diversification, Size, and Risk at Bank Holding Companies. Journal of Money, Credit and Banking 29, 300. https://doi.org/ 10.2307/2953695.
- DeYoung, R., Hunter, W.C., Udell, G.F., 2004. The Past, Present, and Probable Future for Community Banks. Journal of Financial Services Research 25, 85–133. https://doi.org/10.1023/B:FINA.0000020656.65653.79.
- Diallo, B., 2015. Bank competition and crises revisited: New results. Economics Letters 129, 81-86. https://doi.org/10.1016/j.econlet.2015.02.015.
- Dietsch, M., Lozano-Vivas, A., 2000. How the environment determines banking efficiency: A comparison between French and Spanish industries. Journal of Banking & Finance 24, 985–1004. https://doi.org/10.1016/S0378-4266(99)00115-6.
- Enria, A., 2021. Crisis management for medium-sized banks: the case for a European approach. Keynote speech at the Banca d'Italia workshop on the crisis management framework for banks in the EU, Frankfurt am Main, 15 January Available at . https://www.bankingsupervision.europa.eu/press/speeches/date/ 2021/html/ssm.sp210115~e00efc6968.en.html (accessed 26 July 2024).
- Enria, A., 2023. Well-run banks don't fail why governance is an enduring theme in banking crisis. Keynote speech at the 22nd Annual International Conference on Policy Challenges for the Financial Sector organised by the World Bank, International Monetary Fund and Federal Reserve System, Washington, D.C., 1 June. Available at https://www.bankingsupervision.europa.eu/press/speeches/date/2023/html/ssm.sp230601~0d92d89e8c.en.html (accessed 26 July 2024).
- Estrella, A., Park, S., Peristiani, S., 2000. Capital ratios as predictors of bank failure. *Economic policy review*, 6(2). available at https://www.newyorkfed.org/research/epr/00v06n2/0007estr.html (accessed 26 July 2024).
- European Central Bank, 2022. LSI Supervision Report. Available at https://www.bankingsupervision.europa.eu/ecb/pub/html/LSIreport/ssm. LSIreport2022~aac442c1a3.en.html#toc9 (accessed 26 July 2024).
- European Commission, 2023. Impact Assessment Report. Commission Staff Working Document SWD (2023) 225 final. Available at https://eur-lex.europa.eu/ LexUriServ/LexUriServ.do?uri=SWD:2023:0225:FIN:EN:PDF (accessed 26 July 2024).
- Federal Deposit Insurance Corporation, 2020. FDIC Community Banking Study. December available at https://www.fdic.gov/resources/community-banking/report/ 2020/2020-cbi-study-full.pdf (accessed 26 July 2024).
- Fernandez-Bollo, E., Andreeva, D., Grodzicki, M., Handal, L., Portier, R., 2021. Euro area bank profitability and consolidation. Banco de Espana. Financial Stability Review. Issue 40, Spring Available at https://repositorio.bde.es/bitstream/123456789/17322/1/4_Consolidation_FSR.pdf (accessed 26 July 2024). Fiordelisi, F., Molyneux, P., 2006. Shareholder value in banking, Palgrave Macmillan, U.K. https://doi.org/10.1057/9780230595927.
- Flordelis, F., Mayneus, F., Zoot, Sinkerforder vinte in binking, Faigure interimini, Oct. https://doi.org/10.103/9700220050527.
 Flordelis, F., Mare, D.S., 2013. Probability of default and efficiency in cooperative banking. Journal of International Financial Markets, Institutions and Money 26, 30–45. https://doi.org/10.1016/j.intfin.2013.03.003.
- Fiordelisi, F., Grimaldi, S., Lopez, J.S., Mazzilis, M.C., Ricci, O., 2022. The beauty of being involved: The case of cooperative banks. British Journal of Management. 00, 1–23. https://doi.org/10.1111/1467-8551.12698.
- Gardó S., Klaus, B., 2019. Overcapacities in banking: measurements, trends and determinants, European Central Bank, Occasional Paper Series No 236/November available at https://www.ecb.europa.eu/pub/pdf/scpops/ecb.op236~3021bf6dbb.en.pdf?fdf54d42561f8add08b2cac3ab83ccfd (accessed 26 July 2024).
- Gonzales-Hermosillo, B., 1999. Determinants of Ex-Ante Banking System Distress: A Macro-Micro Empirical Exploration of Some Recent Episodes. IMF Working Papers No.1999/033, available at https://www.imf.org/en/Publications/WP/Issues/2016/12/30/Determinants-of-Ex-Ante-Banking-System-Distress-A-Macro-Micro-Empirical-Exploration-of-Some-2908 (accessed 26 July 2024).
- Gržeta, I., Žicovič, S., Žicovič, I.T., 2023. Size matters: analyzing bank profitability and efficiency under Basel III framework. Financial Innovation 9, 43. https://doi. org/10.1186/s40854-022-00412-y.
- Guo, G., 1993. Event-History Analysis for Left-Truncated Data. Sociological Methodology 23, 217. https://doi.org/10.2307/271011.
- Hakenes, H., Schnabel, I., 2011. Bank size and risk-taking under Basel II. Journal of Banking & Finance 35, 1436–1449. https://doi.org/10.1016/j. ibankfin.2010.10.031.
- Halling, M., Hayden, E., 2008. Bank failure prediction: a two-step survival time approach. IFC Bulletins chapters, in: Bank for International Settlements (ed.), The IFC's contribution to the 56th ISI Session, Lisbon, August 2007, volume 28, pages 48-73, Bank for International Settlements. Available at https://www.bis.org/ ifc/oubl/ifcb28.ndf (accessed 26. July 2024).
- Hardy, D., 1998. Are banking crises predictable? Finance and development, December, volume 35, Number 4. International Monetary Fund, Washington, available at https://www.imf.org/external/pubs/ft/fandd/1998/12/hardy.htm.
- Hosmer, D.W., Lemeshow, S., 2000. Applied Logistic Regression, Wiley Series in Probability and Statistics, (2nd Edition). John Wiley & Sons, Inc. 10.1002/0471722146.
- Ioannidis, C., Pasiouras, F., Zopounidis, C., 2010. Assessing bank soundness with classification techniques. Omega 38, 345–357. https://doi.org/10.1016/j. omega.2009.10.009.
- Jenkins, S.P., 2005. Survival Analysis. Unpublished Lecture Notes manuscript, Institute for Social and Economic Research, University of Essex. Available at http:// www.iser.essex.ac.uk/files/teaching/stephenj/ec968/pdfs/ec968lnotesv6.pdf (accessed 26 July 2024).
- Jin, J., Kanagaretnam, K., Lobo, G.J., 2018. Discretion in bank loan loss allowance, risk taking and earnings management. Accounting & Finance 58, 171–193. https://doi.org/10.1111/acfi.12210.

Kanagaretnam, K., Lobo, G.J., Mathieu, R., 2004. Earnings Management to Reduce Earnings Variability: Evidence from Bank Loan Loss Provisions. Review of Accounting and Finance 3, 128–148. https://doi.org/10.1108/eb043399.

Keeley, M.C., 1990. Deposit Insurance, Risk, and Market Power in Banking. The American Economic Review 80, no. 5, 1183–1200. Available at http://www.jstor.org/ stable/2006769 (accessed 26 July 2024).

Kočenda, E., Iwasaki, I., 2022. Bank survival around the world: A meta-analytic review. Journal of Economic Surveys 36, 108–156. https://doi.org/10.1111/ joes.12451.

Li, L., Strahan, P.E., 2021. Who Supplies PPP Loans (and Does It Matter)? Banks, Relationships, and the COVID Crisis. J. Financ. Quant. Anal. 56, 2411–2438. https:// doi.org/10.1017/S0022109021000405.

Liu, H., Molyneux, P., Wilson, J.O.S., 2013. Competition and Stability in European Banking: a Regional Analysis. The Manchester School 81, 176–201. https://doi.org/10.1111/j.1467-9957.2011.02285.x.

Lo Duca, M., Peltonen, T., 2013. Assessing systemic risks and predicting systemic events. Journal of Banking & Finance 37 (7), 2183–2195. https://doi.org/10.1016/j.jbankfin.2012.06.010.

Lopez, J.A., Rose, A.K., Spiegel, M.M., 2020. Why have negative nominal interest rates had such a small effect on bank performance? Cross Country Evidence. European Economic Review 124. https://doi.org/10.1016/j.euroecorev.2020.103402.

Maddaloni, A., Peydró, J.-L. 2013. Monetary policy, macroprudential policy, and banking stability: evidence from the euro area, International Journal of Central Banking, vol. 9, no. 1, 121–69 Available at https://www.ijcb.org/journal/ijcb13q1a6.pdf (accessed 26 July 2024).

Maggiolini, P., Mistrulli, P.E., 2005. A survival analysis of de novo co-operative credit banks. Empirical Economics 30, 359–378. https://doi.org/10.1007/s00181-005-0237-9.

Männasoo, K., Mayes, D.G., 2009. Explaining bank distress in Eastern European transition economies. Journal of Banking & Finance 33, 244–253. https://doi.org/ 10.1016/j.jbankfin.2008.07.016.

Marcus, A.J., 1984. Deregulation and bank financial policy. Journal of Banking & Finance 8, 557–565. https://doi.org/10.1016/S0378-4266(84)80046-1. Mare, D.S., 2015. Contribution of macroeconomic factors to the prediction of small bank failures. Journal of International Financial Markets, Institutions and Money

39, 25–39. https://doi.org/10.1016/j.intfin.2015.05.005. McFadden, D., 2008. Human Capital Accumulation and Depreciation. Review of Agricultural Economics 30, 379–385. https://doi.org/10.1111/j.1467-

9353.2008.00411.x.

Mercieca, S., Schaeck, K., Wolfe, S., 2007. Small European banks: Benefits from diversification? Journal of Banking & Finance 31, 1975–1998. https://doi.org/ 10.1016/j.jbankfin.2007.01.004.

Molyneux, P., Reghezza, A., Xie, R., 2019. Bank margins and profits in a world of negative rates. Journal of Banking & Finance 107, 105613. https://doi.org/10.1016/j.jbankfin.2019.105613.

Oecd, 2021. Education at a Glance 2021: OECD Indicators. OECD Publishing, Paris, https://doi.org/10.1787/b35a14e5-en.

Ostergaard, C., Schindele, I., Vale, B., 2016. Social Capital and the Viability of Stakeholder-Oriented Firms: Evidence from Savings Banks*. Review of Finance 20, 1673–1718. https://doi.org/10.1093/rof/rfv047.

Pappas, V., Ongena, S., Izzeldin, M., Fuertes, A.M., 2017. A Survival Analysis of Islamic and Conventional Banks. Journal of Financial Services Research 51, 221–256. https://doi.org/10.1007/s10693-016-0239-0.

Petach, L., Weiler, S., Conroy, T., 2021. It's a wonderful loan: local financial composition, community banks, and economic resilience. Journal of Banking & Finance 126, 106077. https://doi.org/10.1016/j.jbankfin.2021.106077.

Poghosyan, T., Čihak, M., 2011. Determinants of Bank Distress in Europe: Evidence from a New Data Set. J Financ Serv Res 40, 163–184. https://doi.org/10.1007/ s10693-011-0103-1.

Prentice, R.L., Gloeckler, L.A., 1978. Regression Analysis of Grouped Survival Data with Application to Breast Cancer Data. Biometrics 34, 57. https://doi.org/ 10.2307/2529588.

Rasmusen, E., 1988. Mutual banks and stock banks. Journal of Law and Economics 31, 395-421. https://doi.org/10.1086/467162.

Roy, A.D., 1952. Safety First and the Holding of Assets. Econometrica 20, 431. https://doi.org/10.2307/1907413.

Sarlin, P., Peltonen, T., 2013. Mapping the state of financial stability. Journal of International Financial Markets, Institutions & Money 26, 46–76. https://doi.org/ 10.1016/j.intfin.2013.05.002.

Schaeck, K., Čihak, M., Wolfe, S., 2009. Are Competitive Banking Systems More Stable? Journal of Money, Credit and Banking 41, 711–734. https://doi.org/10.1111/ j.1538-4616.2009.00228.x.

Scott Langford, W., Feldman, M.P., 2022. We miss you George Bailey: the effect of local banking conditions on the county-level timing of the Great Recession. Regional Studies 57, 1832–1850. https://doi.org/10.1080/00343404.2022.2141703.

Shim, J., 2013. Bank capital buffer and portfolio risk: The influence of business cycle and revenue diversification. Journal of Banking & Finance 37, 761–772. https://doi.org/10.1016/j.jbankfin.2012.10.002.

Single Resolution Board, 2023. Small and Medium-Sized Banks: Resolution Planning and Crisis Management Report for Less Significant Institutions in 2022 and 2023, October. Available at https://www.srb.europa.eu/system/files/media/document/2023-10-03_LSI-Small-and-medium-sized-banks-2022-2023.pdf (accessed 26 July 2024).

Stigler, G.J., 1958. The economies of scale. The Journal of Law and Economics 1 (1), 54-71. https://doi.org/10.1086/466541.

Stiroh, K.J., 2004. Do Community Banks Benefit from Diversification? Journal of Financial Services Research 25, 135–160. https://doi.org/10.1023/B: FINA 0000020657 59334 76

Stiroh, K.J., Rumble, A., 2006. The dark side of diversification: The case of US financial holding companies. Journal of Banking and Finance 30, 2131–2161. https://doi.org/10.1016/j.jbankfin.2005.04.030.

Strobel, F., 2011. Bank insolvency risk and Z -score measures with unimodal returns. Applied Economics Letters 18, 1683–1685. https://doi.org/10.1080/ 13504851.2011.558474.

Taylor, J.B., 1993. Discretion versus policy rules in practice. Carnegie-Rochester Conference Series on Public Policy 39, 195–214. https://doi.org/10.1016/0167-2231 (93)90009-L.

Tran, D.V., Hassan, M.K., Houston, R., 2020. Discretionary loan loss provision behavior in the US banking industry. Rev Quant Finan Acc 55, 605–645. https://doi.org/10.1007/s11156-019-00854-z.

Turk Ariss, R., 2010. On the implications of market power in banking: Evidence from developing countries. Journal of Banking & Finance 34, 765–775. https://doi.org/10.1016/j.jbankfin.2009.09.004.

Vazquez F., Federico P., 2012. Bank Funding Structures and Risk: Evidence From the Global Financial Crisis. IMF Working Paper WP/12/29, January. Available at https://www.imf.org/en/Publications/WP/Issues/2016/12/31/Bank-Funding-Structures-and-Risk-Evidence-From-the-Global-Financial-Crisis-25676 (accessed 26 July 2024).

Wheelock, D.C., Wilson, P.W., 2000. Why do banks disappear? The determinants of US bank failures and acquisitions. Review of Economics and Statistics 82 (1), 127–138. https://www.jstor.org/stable/2646678.

Wooldridge, J.M., 2010. Econometric Analysis of Cross Section and Panel Data. MIT Press, MA.

Worthington, A.C., 2004. Determinants of merger and acquisition activity in Australian cooperative deposit-taking institutions. Journal of Business Research 57 (1), 47–57. https://doi.org/10.1016/S0148-2963(02)00283-7.

Yeager, T.J., 2004. The demise of community banks? Local economic shocks are not to blame. Journal of Banking & Finance 28, 2135–2153. https://doi.org/10.1016/j.jbankfin.2003.08.004.